HAZARDOUS TIMES FOR MONETARY POLICY: WHAT DO TWENTY-THREE MILLION BANK LOANS SAY ABOUT THE EFFECTS OF MONETARY POLICY ON CREDIT RISK-TAKING?

2008

Gabriel Jiménez, Steven Ongena, José Luis Peydró and Jesús Saurina

Documentos de Trabajo N.º 0833

BANCODEESPAÑA

Eurosistema

HAZARDOUS TIMES FOR MONETARY POLICY: WHAT DO TWENTY-THREE MILLION BANK LOANS SAY ABOUT THE EFFECTS OF MONETARY POLICY ON CREDIT RISK-TAKING?

HAZARDOUS TIMES FOR MONETARY POLICY: WHAT DO TWENTY-THREE MILLION BANK LOANS SAY ABOUT THE EFFECTS OF MONETARY POLICY ON CREDIT RISK-TAKING?⁽¹⁾

Gabriel Jiménez

BANCO DE ESPAÑA

Steven Ongena

CENTER-TILBURG UNIVERSITY AND CEPR

José Luis Peydró

EUROPEAN CENTRAL BANK

Jesús Saurina

BANCO DE ESPAÑA

(*) We are grateful to Sigbjørn Atle Berg, Sudipto Bhattacharya, Lamont Black, José Manuel Campa, Mark Carey, Stephen Cecchetti, Wouter den Haan, Enrica Detragiache, Giovanni Dell'Ariccia, Valeriya Dinger, Philipp Hartmann, Debbie Lucas, Frank Smets, participants in the Annual Conference on Bank Structure and Competition on "Credit Market Turmoil: Causes, Consequences, and Cures" (Chicago), European Central Bank Conference on "The Implications of Changes in Banking and Financing on the Monetary Policy Transmission Mechanism" (Frankfurt), the International Financial Research Forum on "Financial Risks" (Paris), the International Monetary Fund Conference on "Financial Cycles, Liquidity and Securitization" (Washington DC), the Financial Intermediation Research Society Conference (Anchorage), the Unicredit Group Banking and Finance Conference on the "Span and Scope of Banks, Stability and Regulation" (Naples), and the Workshop on "Banking and Financial Intermediation" (Rimini), and seminar participants at the Banco central do Brasil, Bank of England, the Board of Governors of the Federal Reserve System, the Bundesbank, CEMFI, the European Central Bank, the Federal Reserve Banks of Philadelphia and New York, the International Monetary Fund, and the Universities of Amsterdam, Bonn, Frankfurt and Warwick for helpful comments. We thank Marco lo Duca for excellent research assistance. We also thank the suggestions of an anonymous referee that have substantially improved the paper. Ongena acknowledges the hospitality of the European Central Bank and the Swiss National Bank while writing this paper. Any views expressed are only those of the authors and should not be attributed to the Banco de *España*, the European Central Bank or the European Central Bank or the European Central Bank or the European Central Bank and the Swiss National Bank while writing this paper. Any views expressed are only those of the authors and should not be attributed to the *Banco de España*, the European Central Bank or the European Central Bank

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the INTERNET at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2008

ISSN: 0213-2710 (print) ISSN: 1579-8666 (on line) Depósito legal: Unidad de Publicaciones, Banco de España

Abstract

We identify the impact of short-term interest rates on credit risk-taking by analyzing a comprehensive credit register from Spain, a country where for the last twenty years monetary policy was mostly decided abroad. Discrete choice, within borrower comparison and duration analyses show that lower overnight rates prior to loan origination lead banks to lend more to borrowers with a worse credit history and to grant more loans with a higher per-period probability of default. Lower overnight rates during the life of the loan reduce this probability. Bank, borrower and market characteristics determine the impact of overnight rates on credit risk-taking.

Keywords: monetary policy, low interest rates, financial stability, lending standards, credit risk-taking, credit composition, business cycle, liquidity risk.

JEL: E44, E5, G21.

1 Introduction

Problems in the credit markets first surfaced in the summer of 2007. Since then liquidity recurrently evaporated almost entirely from the interbank markets and central banks intervened worldwide on a scale not often seen before. Large reputable banks had to write down substantial portions of their loan portfolios. Market commentators immediately argued that during the long period of very low interest rates, stretching from 2001 to 2005, banks had softened their lending standards and taken on excessive credit risk. But, at the same time, market participants started clamoring for central banks to reduce interest rates again to alleviate their financial predicament. Hazardous times for monetary policy indeed.¹

But do low short-term interest rates really spur credit risk-taking by banks and are their subsequent sharp increases one of the contributing factors to the current crisis? To answer these questions we study how short-term interest rates influence credit risk-taking by banks using a unique and comprehensive dataset containing bank loan contract information from Spain. Monetary policy was fairly exogenous during the last twenty years in Spain and was mostly "decided in Frankfurt". Hence, we study how the German and then Euro overnight interest rates affected the riskiness of bank loans.

While the impact of changes in the short-term interest rates on the aggregate volume of credit in the economy has been widely analyzed [Bernanke and Blinder (1992), Bernanke and Gertler (1995) and Kashyap and Stein (2000) among others], this paper is the first to empirically study its impact on the banks' appetite for credit risk. Recent theoretical work shows how changes in short-term interest rates may affect *credit risk-taking* by financial institutions. As the change in the demand for investment through the interest rate channel may also affect the marginal borrower's riskiness, disentangling demand from supply of credit is one of the key challenges addressed by our empirical contribution.

Lower interest rates —by improving borrowers' net worth— may result in banks to lend to borrowers that were deemed in the past to be too risky [Bernanke, Gertler and Gilchrist (1996)] or to lend to borrowers with fewer pledgeable assets [Matsuyama (2007)]. In the "balance sheet channel" lenders embrace borrowers that were deemed too risky *in the past* but have become in effect less risky due to the lower interest rates. Yet lower interest rates may push financiers beyond this category of 'redeemed borrowers' to finance firms and projects that are actually riskier *in the present*, i.e., to grant loans with a longer maturity (and more liquidity risk) and even a higher per period probability of default (and more *pure* credit risk). This latter effect of low rates has been labeled the "credit risk-taking channel" of monetary policy [following Borio and Zhu (2007)] and can be considered part of the credit channel.

Lower interest rates may reduce the threat of deposit withdrawals [Diamond and Rajan (2006)], abate adverse selection problems in credit markets [Dell'Ariccia and Marquez (2006)] or improve banks' net worth [Stiglitz and Greenwald (2003)], for example, allowing

^{1.} Nominal rates were the lowest in almost four decades and below Taylor rates in many countries while real rates were negative [see Ahrend, Cournède and Price (2008), Taylor (2007) and multiple editorials and op-eds in *The Wall Street Journal, The Financial Times* and *The Economist* for example]. Loan securitization may have intensified risk-taking [Keys, Mukherjee, Seru and Vig (2008), Mian and Sufi (2008)]. Expansionary monetary policy and credit risk-taking followed by restrictive monetary policy possibly led to the financial crisis during the 1990s in Japan [see Allen and Gale (2004) for example].

banks to relax their lending standards and to increase their credit risk-taking. Low levels of short-term interest rates may further make riskless assets less attractive for financial institutions and lead to a search-for-yield [Rajan (2006)]. On the other hand, higher interest rates increase the opportunity costs for banks to hold cash thus making risky alternatives more attractive [Smith (2002)], or may reduce the banks' net worth or charter value enough to make a "gambling for resurrection" strategy attractive [Kane (1989) and Hellman, Murdock and Stiglitz (2000)], thus making the impact of short-term interest rates on credit risk-taking ultimately a critical yet unaddressed empirical question.

Motivated by these theoretical developments we study the impact of monetary policy on the credit risk-taking by a particular group of financiers, i.e., banks. Banks may not only arise to overcome key informational and contractual problems in modern theoretical work [as in Diamond (1984) for example], but are also the main providers of credit in most economies (Source: *International Finance Statistics* of the *International Monetary Fund*). Credit risk may be the most important risk type banks face [Kuritzkes and Schuermann (2008)].

An empirical investigation of interest rates and bank risk-taking may be warranted, but it presents steep identification challenges. Comprehensive panel data on individual bank loans is needed to dynamically account for changing loan conditions, to adequately assess credit risk, and to disentangle the supply of credit to risky borrowers from its demand. A fairly exogenous monetary policy and ample controls for other important macroeconomic factors (the growth in real gross domestic product, the inflation rate and a measure of country risk for example) are also crucial for econometric identification.

The Credit Register of the Banco de España (CIR) is uniquely suited to fulfill these many requirements. The CIR records detailed monthly information on *all*, new and outstanding, commercial/industrial and financial loans (over 6,000 Euros) to non-financial firms by *all* credit institutions in Spain during the last twenty-three years —generating almost twenty-three million bank loan records in total. Its comprehensiveness directly averts any concerns about unobserved changes in bank lending. The CIR contains loan conditions and performance variables that are essential to our analysis, and tracks key borrower and bank characteristics, including borrower and bank identity.

Banks continue to play a key role in the Spanish economy and in the financing of the corporate sector. In 2006 for example their deposits (credits) to GDP equaled 132% (164%). Most non-financial firms had no access to bond financing and the securitization of commercial and industrial loans is still very low (4.8% in 2006).

Spain formally joined the European Monetary Mechanism in 1989; it had implicitly been part since 1988, after joining the European Union in 1986. Monetary conditions consequently became fairly exogenous and basically "set in Frankfurt", first through the fixed exchange rate policy with the *Deutsche Mark* and as of January 1, 1999 within the Eurosystem. Consequently, we can use the German then Euro overnight interbank rates as an exogenous measure of the stance of monetary policy (alternatively, we instrument the Spanish by the German overnight rate, a strong instrument as our later reported results clearly indicate). In addition to the fairly exogenous monetary conditions and, as both the *Deutsche Bundesbank* and the European Central Bank (ECB) were mandated to principally maintain price stability, controlling for other macroeconomic conditions should allay concerns of reverse causality and omitted variables' bias.

Using discrete choice models, within borrower comparison models and duration analyses, we investigate whether short-term interest rates prior to loan origination influence credit risk-taking by banks. One of our main challenges is to disentangle demand from supply of risky credit and to account for potential changes in credit quality of the marginal borrower: When overnight rates are lower, quality possibly worsens as more —and more risky— projects surpass a zero net present value hurdle for example.

We introduce borrower characteristics, or identity, and loan conditions. The former set of controls absorbs the changes in the pool of borrowers, the latter the adjustments banks make trying to keep their level of risk-taking constant (if banks want to keep their credit risk constant they will tighten lending standards when they face a riskier pool of borrowers or projects). We employ within borrower comparisons for further identification. Borrowers connected to multiple banks are grouped as either risky or safe and we study the proportion of their financing provided by banks with a high or low appetite (and/or ability) for credit risk when interest rates change.

Our findings are both robust and economically relevant. Lower interest rates prior to the origination of the loans precede more lending to borrowers with either a *bad* or *no* credit history. More importantly, banks also grant more loans with a substantially higher *hazard rate* (which is a default probability normalized per time; a normalization that is desirable as loan maturity may also be affected by overnight rates). In striking contrast, once loans are outstanding lower interest rates imply lower hazard rates conform to standard theoretical predictions.

In sum, when monetary policy is expansive, not only do banks give more loans to borrowers with either a bad or no credit history, but also the new loans themselves are more hazardous. Consequently, better borrowers' net worth and a higher appetite for liquidity risk are not the only motives for the banks' new engagements. Following a monetary expansion banks also want to take more credit risk. These main findings are robust to many alterations and for example similar in the *Bundesbank* and ECB sub-periods (i.e., before and after 1999).

We also find robust evidence, confirming theoretical predictions, that smaller banks' incentives and ability for risk-taking are more affected by the stance of monetary policy than larger banks' incentives. The effect of the stance of monetary policy on credit risk-taking further depends on bank liquidity and ownership type, and on the level of banking competition and new borrower entry in the local area. All in all, our results suggest that the stance of monetary policy influences banks' appetite for credit risk.

Higher GDP growth reduces lending to borrowers with bad credit history, and lowers credit risk on new *and* outstanding loans. We note that both lower short-term interest rates and higher GDP growth imply higher borrower net worth and, therefore, fewer agency problems between lenders and borrowers. However, the effect of GDP growth on credit risk-taking is different from the effect of short-term interest rates. This implies that there may be other financial inefficiencies [Rajan (2006) and Diamond and Rajan (2006)] that explain the results of this paper. In addition, higher inflation during the life of the loan reduces credit risk, maybe because it cuts the real value of debt [Allen and Gale (2004) and Diamond and Rajan (2006)]. In contrast higher inflation before loan origination implies higher risk taking. Finally, long-term interest rates have a weaker impact on credit risk-taking than short-term rates, possible because banks themselves rely mainly on short-term debt [Diamond and Rajan (2006)].

To the best of our knowledge, loannidou, Ongena and Peydró (2007) and this paper are the first to investigate the impact of monetary policy on the credit risk-taking by banks.² loannidou, Ongena and Peydró (2007) use the Credit Register from Bolivia —where the banking system is almost completely dollarized— from 1999 to 2003. When the federal funds rate decreases, they find, not only do banks take more risk but they also reduce their loan spreads. Despite using credit registers covering different countries, time periods, and monetary policy regimes, both papers find strikingly similar results.

Consequently these results have important policy implications regarding the link between monetary policy and financial stability, and regarding the root causes of the turbulences in credit markets since the summer of 2007. The moment of highest credit risk, our results suggest, is when short-term rates that were very low for a long time period suddenly increase steeply: At that moment, the net worth of an already risky pool of borrowers is suddenly eroded possibly leading to non-performance and defaults. Our results also suggest a way forward: Reducing interest rates again lowers the credit risk of all outstanding loans, making banks more willing to again accept credit risk thereby reducing the tensions in the credit markets.

The rest of the paper proceeds as follows. Section 2 details and explains the testable predictions derived from theory. Section 3 reviews our empirical strategy and introduces the data and the variables employed in our empirical specifications. Section 4 discusses the results on borrower selection, within borrower comparison, and time to default of bank loans. Section 5 summarizes the results, discusses policy implications and concludes.

^{2.} Dell'Ariccia, Igan and Laeven (2008) "provide hints" (sic) on the potential effects of monetary policy on banks' risk-taking. In line with our findings, their results are consistent with the idea that low interest rates in the U.S. may have loosened lending standards both directly and through their effect on real estate prices. Den Haan, Sumner and Yamashiro (2007) find that restrictive monetary policy reduces consumer and real estate lending in particular and argue that high short-term rates could imply a decline in bank risk-taking. Gertler and Gilchrist (1993), Gertler and Gilchrist (1994) and other papers documenting the strength of the balance sheet channel show that contractionary monetary policy results in less bank lending to small firms, findings that are consistent not only with lower borrower net worth but also with less bank risk-taking. Indeed Black and Rosen (2008) show that a lowering of the federal funds rate lengthens loan maturity and reallocates lending from large to small firms. And in a different setting Bernanke and Kuttner (2005) find that higher unanticipated interest rates reduce equity prices. One of their interpretations of this finding is that tight money may reduce the willingness of stock investors to bear risk. Rigobon and Sack (2004) show that higher interest rates reduce equily prices, especially on NASDAQ where arguably more risky firms are listed.

2 Literature Review and Testable Implications

2.1 Credit Channel of Monetary Policy

Does the stance of monetary policy, short-term interest rates in particular,³ affect the appetite for credit risk of financial intermediaries? In addressing this question, that is key to our investigation, one needs to recognize a large literature in economics that has investigated whether and exactly how the conduct of monetary policy has effects for the real economy.

Short-term interest rates may affect investment and hence firms' demand for credit (the classical interest rate channel), while at the same time short-term interest rates may affect credit markets and the banking system in particular [for the credit channel of monetary policy transmission, see Bernanke and Gertler (1995)].⁴ Because of credit market imperfections —asymmetries of information and incompleteness of contracts— contractive monetary policy may reduce the ability and incentive of banks to supply loans [Bernanke and Blinder (1988), Bernanke and Gertler (1989), Lang and Nakamura (1995), Gertler and Gilchrist (1994), among others]. Bernanke and Blinder (1992) indeed find that significant movements in bank aggregate lending volume follow changes in the stance of monetary policy.⁵ To control for loan demand, Kashyap and Stein (2000) analyze whether there are also important cross-sectional differences in the way that banks respond to monetary policy shocks. They find that illiquid banks respond most and that this variation can be attributed to bank size.

Higher levels of interest rates directly and/or indirectly reduce borrowers' net worth [Bernanke and Gertler (1995)]. This reduction in net worth worsens the agency problems between banks and their borrowers in turn making banks fly to quality [Bernanke, Gertler and Gilchrist (1996) and Bernanke, Gertler and Gilchrist (1999)]. In Matsuyama (2007) borrowers are heterogeneous in productivity and asset pledgeability, and banks change their optimal *credit composition* among borrowers according to the business cycle —in fact, there is a change in credit composition without a change in bank volume. Lower short-term rates for example increase borrowers' net worth making banks more willing to lend to borrowers with fewer pledgeable assets.

2.2 Credit Risk-Taking Channel of Monetary Policy

All theories discussed so far show that expansive monetary policy may increase the volume of loan supply, in particular to borrowers that in the past looked too risky (borrowers with a bad credit history for example that due to an improvement in their net worth are not so risky anymore). However, none of these theories necessarily imply higher risk-taking by the banks, i.e., granting loans with a higher default probability.

In Diamond and Rajan (2006) banks take more risk when monetary policy is expansive. In their model, which provides "a liquidity version of the lending channel" of the monetary policy transmission mechanism, banks finance illiquid long-term projects with liquid

^{3.} In Bernanke and Blinder (1992) and Christiano, Eichenbaum and Evans (1996), among others, the overnight interest rate is an indicator for the stance of monetary policy. The ECB targets the overnight rate as a measure of the stance of its monetary policy.

^{4.} The credit channel comprises both a balance sheet and a bank-lending channel. The latter channel can also be viewed as a balance sheet channel for banks (see Bernanke (2007) for a recent review of this literature).

^{5.} Den Haan, Sumner and Yamashiro (2007) find that monetary policy differently affects consumer, real estate, and business lending by banks. In fact, contractive monetary policy does not decrease the volume of business loans for example. Their findings may be caused by a decline in bank risk-taking when short-term interest rates are high.

demand deposits. This mismatch makes banks reluctant to grant risky loans in times of liquidity shortages. Depending on the aggregate liquidity conditions, monetary intervention can play a useful role by limiting the depositors' incentives to withdraw. Banks will respond by continuing, rather than curtailing, risky credit. Consequently, the stance of monetary policy may affect the banks' appetite for risk.⁶

In Dell'Ariccia and Marquez (2006) banks' incentives to screen depend on their cost of financing, which in turn is determined by short-term interest rates. When there is a monetary tightening, to recover their losses on the "loans to lemons" when not screening, banks have to increase the average loan rate more than the observed increase in the policy rate. This pass-through is clearly larger than if banks were screening. Consequently, if interest rates decrease, the banks' incentives to screen borrowers and to fly to quality (with a lower default probability) wane. Banks also lower their screening standards when there are many loan applicants that are new to *all* banks. A decline in the cost of funds in Sengupta (2007) likewise facilitates entry of outside banks into "high-risk" credit markets, as inclusion of non-credit worthy borrowers in their loan portfolio becomes possible. And in Ruckes (2004) improvements in economic outlook and declines in the average default probability of the borrowers lowers the lenders' screening activity, intensifies price competition and boosts lending to low quality borrowers.

In general, because of information, contract and competition imperfections, banks may act risk aversely, and the degree of their risk aversion may depend not only on their borrowers' but also on their own net worth, Stiglitz and Greenwald (2003) argues. Lower interest rates increase the value of the banks' portfolio of securities and loans, thereby raising banks' net worth and capital. This in turn increases their ability and incentives to take credit risk [see also Stiglitz (2001)].⁷

Rajan (2006) discusses whether and how the stance of monetary policy may affect the risk-appetite of financial intermediaries.⁸ Because of incentives, contractual and informational imperfections, managers of financial intermediaries may take excessive risk when interest rates are low. Financial intermediaries may increase their exposure to liquidity risk for example by exploiting their duration mismatch and by lending more to risky long-term projects. This enhanced risk-taking yields extra returns during normal times. However the extra risk becomes problematic when monetary policy tightens and liquidity evaporates.⁹

^{6.} In Diamond and Rajan (2006) an entrepreneur gets either an early or a late payoff that is fixed (and known). Introducing uncertainty about the payoff level will influence the amount lent to the entrepreneur, we conjecture, without altering the main results of the model. In consequence, monetary policy may affect both the level of liquidity and credit risk taken by the banks.

^{7.} Because of equity rationing, the shocks to the banks' net worth may not be immediately reversible, explaining their potentially large adverse macro-economic consequences. Lower interest rates may also reduce moral hazard and adverse selection problems in credit markets, thereby lessening credit rationing [Stiglitz and Weiss (1981)].

^{8.} Lower policy rates may for example reduce the loan-deposit rate spread, shrink the financial intermediation margin and whet bankers' incentive to take risk to meet some profitability target (*European Central Bank Financial Stability Review*, December 2007). See also Caballero (2006).

^{9.} Ahrend, Cournède and Price (2008), Taylor (2007) and Shiller (2007) discuss excess risk-taking in general. Borio (2003) and Borio and Lowe (2002) assert that monetary policy narrowly focused on controlling short-run goods price inflation is less likely to exert control over credit expansions and asset price inflation (followed by subsequent busts). The increase in the number of booms and busts in recent years is thus, in part, a corollary to "the death of inflation". Borio and Zhu (2007) maintain that recent changes in banking regulation and the financial system may have amplified the impact of monetary policy on the risk-taking by financial intermediaries. Adalid and Detken (2007) find a correspondence between liquidity shocks and aggregate asset prices during asset price boom or bust episodes for 18 OECD countries since the 1970s. Kyotaki and Moore (1997) show how falling interest rates and rising asset prices cause a lending boom by increasing collateral values. As asset prices and collateral values decrease,

All in all, lower short-term interest rates may increase the banks' incentives to take both more liquidity risk (i.e., more loans with a longer maturity) and credit risk (i.e., loans with a higher probability of default). To analyze credit risk-taking controlling for liquidity risk therefore requires investigating the probability of loan default normalized per time period.

2.3 Four Testable Hypotheses

To summarize, there are four testable predictions arising from the theory discussed so far:

(H1) Lower short-term interest rates lead to more lending to borrowers that in the past were riskier but that, because of their higher net worth for example, are now worth engaging.

(H2) Lower interest rates increase the credit risk-taking by banks, i.e., banks not only engage borrowers that are observably riskier *ex ante* but also borrowers with a higher probability of default per time period.

(H3) Default rates are at their highest level when —after a period of low interest rates which boosted credit risk-taking— a significant monetary tightening devalues the net worth of the (outstanding) risky borrowers.

(H4) Not all types of banks are equally affected. Smaller banks for example —because of their lower net worth, lack of diversification and difficulties accessing liquidity during tighter times— whet their appetite for risk more in response to an expansionary monetary policy.

Yet, it is important to note that not all banking theory implies that higher short-term rates go hand-in-hand with decreased credit risk-taking. Higher interest rates increase the opportunity costs for banks to hold cash (held as an insurance against high deposit withdrawals), making risky alternatives more attractive [Smith (2002)]. Higher interest rates may also reduce banks' net worth or charter value, driving banks towards a "gambling for resurrection" strategy [Kane (1989) and Hellman, Murdock and Stiglitz (2000)].¹⁰ Given these conflicting theoretical predictions, determining the impact of monetary policy on bank risk-taking is ultimately a non-trivial, but mostly unaddressed, empirical question.

loan defaults occur. Finally, Allen and Gale (2004) point out that while monetary policy may influence risk taking and hasten the creation of a bubble, it may also help solve credit problems once the bubble bursts. **10.** See also Bhattacharya (1982), Matutes and Vives (2000) and Cordella and Levy-Yeyati (2003).

3 Empirical Strategy

Essential ingredients to econometrically identify the impact of monetary policy stance on credit risk-taking are: (1) A disaggregated and comprehensive dataset on bank loans; (2) a measure of bank credit risk-taking that reflects the loan default risk per period of time; (3) a number of steps to disentangle supply of risky credit from its demand (employing loan and borrower characteristics/identity and interactions with bank characteristics), and (4) an exogenous stance of monetary policy. Spain delivers all of them as the *Credit Register* of the *Banco de España* (CIR) contains comprehensive information on Spanish bank lending over the last twenty-three years, a period during which Spain had a reasonably exogenous monetary policy.

3.1 Disaggregated and Comprehensive Dataset on Bank Loans

We aim to study the impact of the stance of monetary policy on the credit risk-taking of banks. We are therefore in the first place not interested in the impact of monetary policy on the risk of total outstanding loan portfolios, but want to focus on the credit risk of new loans being granted.

The CIR, first employed by Jiménez and Saurina (2004) and Jiménez, Salas and Saurina (2006) contains confidential and very detailed information at the loan level on (almost) *all* loans granted by *all* credit institutions operating in Spain during a twenty-three year period. The CIR is *almost* comprehensive, as the reporting threshold for a loan is only 6,000 Euros. This low threshold alleviates any concerns about unobservable changes in bank credit to small and medium sized enterprises, which may be influenced by changes in monetary policy.¹¹ We have borrower, bank and loan information. For the *borrower* we know its identity, province, industry, bank indebtedness level, credit history and the number of banking relationships; for the *bank* we know its identity, legal status, size, non-performing loan ratio and other bank characteristics; and for the *loan* we know the type of instrument, currency, maturity, degree of collateralization, default status, and amount.

We do not have information on the interest rate of the loan. However, when we control for an individual bank risk premium, the empirical results we report are unaffected.¹² Moreover, Jiménez, Salas and Saurina (2006) show that collateral requirements are softened during expansions, while Delgado, Salas and Saurina (2007) find that average loan size (in real terms) and loan maturity increase in upturns. Therefore, it seems reasonable to expect a decline in the loan risk premium during expansions, given that the other risk control tools present in the loans are also relaxed.

Our sample consists of new business loans, granted to non-financial firms by commercial, savings banks and credit cooperatives (the 95% of all the Spanish

^{11.} See Gertler and Gilchrist (1993), Gertler and Gilchrist (1994), Bernanke and Gertler (1995) and Bernanke, Gertler and Gilchrist (1996) for example. Incomplete coverage of the widely used U.S. (National) Survey of Small Business Finances or the more recent Loan Pricing Corporation datasets [e.g., Petersen and Rajan (1994), Berger and Udell (1995), Bharath, Dahiya, Saunders and Srinivasan (2006), and Calomiris and Pornrojnangkool (2006)] may therefore complicate any analysis of bank risk-taking. The reporting threshold in the *Deutsche Bundesbank*'s Credit Register dealt with in Ongena, Tümer-Alkan and von Westernhagen (2007) is 1,500,000 Euros. Other than the Bolivian register employed in Ioannidou and Ongena (2008) and Ioannidou, Ongena and Peydró (2007), we know of no comprehensive credit register (with no or low reporting threshold) that contains borrower identity and individual loan performance.

^{12.} As in Martin-Oliver, Salas and Saurina (2006) we calculate a risk premium for each individual bank and subtract the average risk premium of all banks in that quarter.

financial system), where non-Spanish subsidiaries and branches are excluded. We use quarterly data information from 1984:IV to 2006:IV, which includes at least a complete business cycle in Spain. Every quarter all new loans are selected and tracked over time until they become impaired, repaid or still outstanding at the end of 2006.

We work at the loan level because econometric identification (further discussed below) requires variation in both borrower characteristics and loan conditions. In addition, Den Haan, Sumner and Yamashiro (2007) document that contractive monetary policy does not necessarily reduce the volume of business lending, while Hernando and Martínez-Pagés (2001) find no bank lending channel operative in Spain. Both findings suggest that focusing on new business loans allows us to determine the changes in loan composition without overlooking the concurrent changes in loan volume across loan categories (despite these priors we control for the growth in total and individual bank loan volume in the robustness subsection).

All in all, we have complete records on almost twenty-three million loans.¹³ In order to keep estimations manageable, we randomly sample 3% of these new loans in the CIR and work with 674,127 loans or 1,987,967 loan-quarter observations. We select firms on the basis of their tax identification number to obtain a random sample.¹⁴ We use the same set of loans throughout the paper, unless otherwise stated.¹⁵

3.2 Measuring Credit Risk

We first measure credit risk-taking with ex ante borrower-level proxies. Employing discrete choice models, we analyze whether a loan is granted to a borrower with a good or bad credit history (depending on default on another loan in the previous 6 months), and alternatively, with or without credit history (in the latter case the borrower has more uncertain cash flows).

As explained in Section II an (almost) ideal measure of credit risk is the probability of default of individual loans that is normalized per time period. Indeed, lower short-term rates may lead to more lending to non-prime borrowers and/or a lengthening of average loan maturity. However, neither action needs to imply higher credit risk-taking *per se*, as the net worth of non-prime borrowers also improved because of the lower short term rates and longer loan maturity entails more liquidity risk but not necessarily more credit risk.

Therefore, to isolate the changes in credit risk-taking, we employ duration models and analyze the impact of short-term interest rates prior to loan origination on the loan hazard rate, which is a loan default probability that is normalized per unit of time. This is possible as our dataset includes unique loan repayment information, i.e., not only *whether* the loan is overdue or not but also *when* it defaults.

We use the observed realizations of the time to default as our main measure of bank risk-taking *ex ante.*¹⁶ However, we do observe all new and outstanding loans over a very

^{13.} The entire database contains more than 32,000,000 loans. We focus on the 22,470,900 commercial and financial loans (80% of total loans), excluding leasing, factoring and other specialized loans, granted by commercial banks, savings banks or credit cooperatives (95% of total credit market).

^{14.} Given the way in which tax identification numbers are assigned, all firms with the same last digit for example would comprise a 10% random sample.

^{15.} We randomly re-sample another 3% of the loans and re-run all specifications. We also sample 6% of the loans and run selected specifications. Results (available upon request) are virtually unchanged and illustrate the robustness of our results to the sampling procedure.

^{16.} Internal or external credit ratings of loans or borrowers are not available. However, these measures are often criticized as coarse and unreliable. Internal credit ratings of banks in the U.S., Germany and Argentina for example are

long time period and can control in our estimations for multiple bank characteristics and time-varying macroeconomic conditions, assuaging any concerns that the realizations of the time to default would systematically differ from the officers' expectations.

In addition, employing duration models, we can not only dynamically control for macroeconomic factors that influence credit risk, but also analyze at each point in time during the life of the loan what impact monetary policy has on its default probability. This possibility further helps in our econometric identification. Indeed, a testable prediction from theory implies that higher interest rates may lead to lower credit risk-taking at origination (as banks' net worth is lower), but results in higher credit risk once the loan is outstanding (as the borrowers' net worth is lower). We can actually test this prediction using duration models —i.e., on a loan-by-loan basis we can analyze how overnight rates before loan origination and how the path of overnight rates after loan origination affect the loan hazard rate.

Other potential measures that one could use as proxies for credit risk-taking are loan rates or collateral requirements. However, it is not clear what the expected impact of the short-term interest rates on loan rates or collateral requirements is.¹⁷ Contractive monetary policy for example would imply higher loan spreads and collateral requirements if the banks' risk-appetite decreases. On the other hand, lower risk-appetite implies banks fleeing to quality, and consequently lower loan spreads and collateral requirements for the better average borrower.

3.3 Disentangling Supply of Credit to Risky Borrowers from its Demand

Detailed information in the CIR database allows us to disentangle the supply of credit to risky borrowers from its demand. The identification problem we face is that, when short-term interest rates change, the loan demand by the risky borrowers (or for risky projects) could be affected differently than the loan demand by the not-so-risky (henceforth, "safe") borrowers. Fortunately for our investigation, this identification problem may not be as severe as in the bank lending literature at large because four different steps assure us a proper identification.

First, lower short-term interest rates in general result in a higher loan demand [Kashyap and Stein (2000)], but not necessarily only from risky borrowers. Moreover, if their demand for loans would increase more (than the demand from the safe borrowers) and if the banks' appetite for risk would remain unaffected, we actually would expect observable lending standards to tighten and not to loosen. Therefore, controlling for lending standards such as collateral, maturity and loan amount therefore mitigates the identification challenge.

Second, we rely on a wide set of borrower characteristics, including borrower identity, to absorb the variation in the risk of the pool of the borrowers over the monetary policy cycle.

Third, as in Kashyap and Stein (2000), we exploit the cross-sectional implications of the sensitivity of bank risk-taking to monetary policy according to the strength of banks'

partly based on subjective, non-financial factors [Treacy and Carey (2000), Grunert, Norden and Weber (2005) and Liberti (2004)]. Loan officers can therefore manipulate the ratings (Hertzberg, Liberti and Paravisini (2007)] and give better ratings when interest rates are low. *External* ratings provided by credit rating agencies were widely blamed during the financial crisis of 2007 as having been uninformative, even deceptive, in the years prior to the crisis.

^{17.} The loan rate as a proxy for risk may additionally suffer from the variation over time in the price of risk. Evidence from equity prices [Bernanke and Kuttner (2005)], bond yields [Manganelli and Wolswijk (2007)], buyout pricing [Axelson, Jenkinson, Strömberg and Weisbach (2007)] and loan rates [loannidou, Ongena and Peydró (2007)] suggest this time variation is common across many financial assets.

balance sheet and moral hazard problems (proxied by bank size for example). At each point in time there are more than two hundred banks operational in Spain providing ample cross-sectional variation.

Finally, as in Khwaja and Mian (2008), we observe that many borrowers have multiple bank connections so that we can perform *within* borrower comparisons. We assess how for each borrower the proportion of financing provided by banks with a higher and lower appetite for risk (i.e., small versus large banks) depends on changes in the overnight interest rate. We complete this assessment by differentiating between riskier and safer borrowers (with and without a recent bad credit history). If following a monetary expansion, riskier borrowers connected to both types of banks start borrowing relatively more (than the safer borrowers) from those banks with a higher ability and incentive to increase risk when short-term rates change, one can argue that these banks offer the better deals or ration less the riskier borrowers. Consequently, it is also the banks that are adjusting their appetite for risk, not only the borrowers that are altering their demand for loans.

To conclude, given the nature of the identification problem we face, the range of borrower and bank characteristics we have access to, and the econometric strategies we mobilize, we surmise that making inferences on whether the stance of monetary policy affects the banks' appetite for credit risk is possible.

3.4 Exogenous Monetary Policy

If the level or variations of short-term interest rates were completely exogenous, the coefficient of the regression of credit risk-taking on short-term interest rates would indicate its causal impact. When monetary policy is not exogenous, there are two problems with econometric identification: (i) Reverse causality (e.g., future higher risk may imply current monetary expansion), and (ii) omitted variables (variables correlated with the stance of monetary policy that can also influence risk-taking). We can deal with these two problems relying on two key features of our empirical setting.

First, for the whole period analyzed, short-term interest rates in Spain were decided mostly in Frankfurt, not in Madrid. Implicitly from mid-1988 and explicitly from mid-1989 when Spain joined the European Monetary System and its exchange rate mechanism, the exchange rate target with the *Deutsche Mark* was the main objective of its monetary policy [Banco_de_España (1997)].¹⁸ From 1999 onwards, Spain joined the Eurosystem as one of twelve countries, representing less than 15% of the total economic activity in this group.

Second, the mandate given to the *Bundesbank* and the ECB was primarily to preserve price stability, not to stimulate economic activity. Nevertheless, we control in the specifications for many macroeconomic variables such as Spanish (and German/ Euro Area) GDP growth and their forecasts, inflation, country risk, yield curve measures and other variables. In addition, we also track bank lending to borrowers with bad or no credit history.

^{18.} In 1986 Spain joined the European Union. Consequently, monetary policy started to pay more attention to the exchange rate and, in particular, to the Peseta / *Deutsche Mark* exchange rate. The monetary policy authorities in this way intended to incorporate more discipline and credibility in their fight against inflation. At the same time capital restrictions were being eliminated. As of mid-1988, Spanish monetary policy was no longer independent from the German monetary policy according to the textbook 'Mundell-Fleming trilemma' [see Blanchard (2006) or Krugman and Obstfeld (2006) for example]. It is therefore not entirely surprising that Boivin, Mojon and Giannoni (2008) finds that a shock to German monetary policy has a stronger impact in Spain than in Germany itself. Spain did devalue its currency three times between 1992 and 1993 and also had temporary credit controls the second half of 1989 and during 1990. In non-reported robustness regressions, we include time dummies for the quarters involved and results do not change significantly.

Indeed, if risk is expected to increase in the future and monetary policy reacts by being expansive today, low interest rates could still correspond to higher loan hazard rates even if banks flee to quality because of the looming crisis. However, observing that banks continue to grant loans to risky borrowers (for instance borrowers with no credit history) implies it is not monetary policy simply reacting to future risk but banks actually seeking it.

Our sample period spans more than a complete domestic business cycle as we extract quarterly loan records running from 1984:IV to 2006:IV to study the impact of monetary policy on bank credit risk-taking starting in 1988:II. We use the German and Euro overnight rates as a measure of the stance of monetary policy. While monetary policy was fairly exogenous, as we argued, and hence this measure is therefore appropriate, we also employ either the Spanish overnight interest rate directly or instrumented by the German rate. In addition, we also run all specifications for the pre-1999 period and for the euro period.

To conclude, given the degree of exogeneity of monetary policy in Spain, the macroeconomic controls and the combination of econometric strategies we employ, we are confident that making inferences on whether the stance of monetary policy affects the banks' appetite for credit risk is possible.

4 Results

Figure 1 summarizes the line-up of the models we will now discuss, including the indexing and timing of the variables. We say a loan *I* by bank *b* is granted to borrower *j* in quarter τ . Let *T* denote the time to maturity or the time to default in case of an overdue repayment; hence, repayment or default would occur in quarter $\tau + T$.

The *probit* models use monetary policy conditions present in the quarter prior to the origination of the loan (at τ -1) to explain the probability that the borrower that obtains the loan (at τ) has a bad or no credit history. The *within borrower comparisons* explain the difference in the change in the proportion of lending by small and large banks, to each individual borrower, between borrowers with and without a recent bad credit history (at τ) employing monetary policy conditions present in the quarter prior (at τ -1). *Duration models* differentiate between monetary policy conditions prior to origination (τ -1) and policy conditions prevailing during the life of the loan, either in the quarter prior to loan repayment or default (τ +T-1) or, in all the periods between the loan origination and the loan repayment or default (τ +t, where t=0, 1, ..., T-1).

Table 1 defines the main dependent (Panel A) and independent (Panel B) variables that are employed in the empirical specifications (including those that are used in the tabulated robustness exercises) as well as their descriptive statistics. As independent variables in the models we include measures of monetary policy conditions and an array of bank, borrower, loan, macroeconomic and provincial characteristics and controls. We measure monetary policy conditions using the quarterly average of nominal German and, from 1999:I onwards, Euro overnight interbank interest rates. We label the monetary policy measure prior to loan origination as INTEREST RATE.

Credit risk-taking and, in general, lending behavior may vary across banks. To control for differences in lending by the banks we introduce bank characteristics (coming from the CIR as well as from the banks' balance sheets). We include LN(TOTAL ASSETS) which equals the logarithm of the total assets of the bank, LIQUIDITY RATIO is the amount of liquid assets held by the bank over total assets, and OWN FUNDS/TOTAL ASSETS is the amount of bank equity over total bank assets.¹⁹ We also control for the level of bank risk (i.e., an indicator of the bank's relative risk appetite and/or ability), measured by BANK NPL_b-NPL, which is the difference between the bank and the other banks' non-performing loan ratios. All bank characteristics are measured prior to the loan origination quarter. We further include the type of bank ownership using dummies that equal one if the bank is a commercial bank LISTED, a SAVINGS BANK, or a CREDIT COOPERATIVE and equal zero otherwise.²⁰ As savings banks and credit cooperatives are not listed, the benchmark category is the non-listed commercial bank.

We include borrower and loan characteristics as the composition of the pool of borrowers and loans may change over time and to disentangle supply from demand effects

^{19.} Very few banks in certain periods record negative equity values. Removing these observations does not alter our results and, for consistency reasons, we decided to retain them.

^{20.} Delgado, Salas and Saurina (2007) explain the main features of the Spanish banking system, focusing in particular on the differences in characteristics and behavior of commercial banks (both listed and non-listed), savings banks and credit cooperatives. All of them compete under the same rules although savings banks do not have shareholders.

(we also employ borrower identity in the duration analysis of individual loans).²¹ We define LN(1+BORROWER BANK DEBT) to equal the logarithm of one plus the total amount of borrower bank debt and LN(1+NUMBER OF BANK RELATIONSHIPS) as the logarithm of one plus the number of bank relationships of the borrower. We do not have direct access to the actual date of registration of the firm, but we know when the firm borrowed for the first time since 1984:IV. The variable LN(2+AGE AS BORROWER) measures the age of the borrower in the CIR and for most firms will be highly correlated with actual age.²² In addition to these borrower characteristics, our specifications include ten Industry dummies and fifty Province dummies.

As loan characteristics we include the log of the total loan amount, LN(SIZE OF THE LOAN) and time-invariant loan dummies that equal one if the loan requires COLLATERAL or is a FINANCIAL CREDIT and equal zero otherwise. Four MATURITY dummies stand for the 0 to 3 month, 3 month to 1 year, 1 to 3 year, and 3 to 5 year maturity classes, the benchmark being long-term loans over 5 years. All borrower characteristics are dated in the quarter prior to the origination of the loan and all loan characteristics are naturally dated in the quarter of the origination itself.

To capture general economic and market developments related to business cycle conditions and country risk, we include the growth in real gross domestic product, GPDG, in the quarter prior to the origination of the loan, and at origination itself the Spanish INFLATION rate, a COUNTRY RISK measure (the spread between the ten year Spanish and German government bond rate), and a TIME TREND and TIME TREND² [as in Kashyap and Stein (2000)].²³

4.1 Borrowers with a Bad or without Credit History

We start analyzing measures of credit risk-taking that proxy for risk at the borrower level. We estimate probit models with two different discrete dependent variables. RECENT BAD CREDIT HISTORY is a dummy that equals one if any loan of the borrower was overdue during the previous six months, and equals zero otherwise, and NO CREDIT HISTORY is a dummy that equals one if the borrower receives its first bank loan ever, and equals zero otherwise (i.e., the borrower has or had another loan). Through the CIR database bank loan officers can find out whether loans of potential borrowers were overdue during the last six months and they also have access to the total exposure of the potential borrower.

Table 2 shows the results. We report the coefficients in the first column and the significance levels represented by stars in the second column.²⁴ In the third column (and in italics between parentheses) we tabulate the robust standard errors that are corrected for clustering at the borrower level.²⁵

^{21.} Introducing borrower fixed effects in the probit models leaves the results unchanged.

^{22.} This variable is therefore by construction left censored but removing it or limiting its backward looking horizon does not alter our results.

^{23.} Replacing the spread with the time-varying International Country Risk Guide index does not alter results. We also include measures of banking system efficiency or credit growth, individual bank credit growth, German / Euro inflation and GDP growth, GDP growth forecasts, the volatility of GDP, yield curve measures and house prices. Results are unaffected and we opted to report the more parsimonious models. We revisit the inclusion of these variables in our duration analysis.

^{24. ***} Significant at 1%, ** significant at 5%, and * significant at 10%. For convenience we also indicate the significance levels of the coefficients in the text.

^{25.} Loan default may be correlated at the borrower level, an issue we revisit in robustness. We alternatively cluster at the bank or quarter level but results are mostly unaffected.

The dependent variable in Model I is RECENT BAD CREDIT HISTORY. The probit regression shows that low interest rates imply that banks grant more loans to borrowers thathad a recent bad credit history, i.e., a non-performance of a loan during the last six months. This is one of our key findings. The possibility that we are observing the evergreening of loans by banks can be discarded; this activity should decrease —not increase— when interest rates are low (we further drop individual loan renewals in robustness later in the paper). The effect of the interest rate on engaging risky borrowers (i.e., borrowers with a recent bad credit history) is also economically relevant. A two-standard deviation cut in interest rates from 5% to 1% for example would increase the probability of lending to a risky borrower by a 16%.

The coefficients on the other variables are mostly as expected. Banks with relatively more than average non-performing loans already on their books, fewer own funds and that are non-listed commercial banks or that are credit cooperatives seem more likely to engage borrowers that were overdue during the last six months. Younger firms with fewer bank relationships are also less likely to have been non-performing recently.²⁶ Observing a lower amount of credit and higher collateral requirements also makes it more likely that the borrower was recently non-performing. This is to be expected if banks adjust loan conditions when demand from those risky borrowers is higher. Not including loan or borrower characteristics or, alternatively, including borrower fixed effects leaves the results unaffected.

On the basis of a commonly shared reading of Spanish monetary history we argued earlier that the German/Euro overnight interest rate is the proper and fairly exogenous monetary policy rate. Nevertheless, we replace the INTEREST RATE by the INTEREST RATE in SPAIN instrumented by the German overnight interest rate. The first stage regression of the INTEREST RATE in SPAIN on the German INTEREST RATE, reported in Table 2B, shows that controlling for Spanish macro conditions the German rate is a strong instrument [Staiger and Stock (1997)], with a t-statistic on its coefficient that is larger than eight. Jointly the variables explain 97% of the variation in the Spanish interest rate (the correlation between the two countries' GDP growth rates is only 36%). The effect of the instrumented Spanish interest rate on the likelihood a risky borrower is granted a loan is, as expected, even larger and statistically significant at the 1% level (Model II in Table 2). A cut in interest rates from 5% to 1% in this case increases the probability of lending to a risky borrower by 21%.

Finally, as in Kashyap and Stein (2000) and using cross-sectional variation in bank characteristics to further identify the supply effect, we add the interaction of the INTEREST RATE with LN(TOTAL ASSETS). Small banks may react more strongly in their risk-taking to changes in the interest rate since they may have worse access to liquidity for example [Diamond and Rajan (2006)]. The coefficient on the interaction term in Model III shows this is indeed the case.

To account for the unlikely possibility that banks evergreen and also for the general interpretation of the results, Models IV to VI show that low interest rates also correspond to banks granting more loans to borrowers that have NO CREDIT HISTORY. New borrowers have in general more uncertain cash flows and are therefore riskier. Results are similar to the results on borrowers with a bad credit history. Note that these results reinforce the

^{26.} Firms with bilateral bank relationships may perform better. Badly performing firms may also initiate more bank relationships [Bhattacharya and Chiesa (1995) and Yosha (1995); for evidence see Degryse and Ongena (2001), Farinha and Santos (2002) and Fok, Chang and Lee (2004), among others]. The result on age is overturned in the duration analysis.

interpretation that lenders take on higher credit risk when short-term interest rates are lower and not that lower interest rates and high credit risk are just a consequence of worse economic fundamentals in the economy.

All in all, banks seem to lend to riskier borrowers when interest rates are low prior to loan origination, and given the controls for GDP growth, inflation, country risk and other variables, the results further suggest that when stance of monetary policy is more expansive, credit risk-taking is higher.²⁷ Therefore, we find support for H1.

4.2 Within Borrower Comparison

Despite the inclusion of borrower characteristics and loan conditions, and the fact that our estimates so far suggest that small banks react more to interest rate changes than large banks, concerns may linger whether our findings pick up supply side effects. We therefore turn to a within borrower comparison to take our additional and final identification step.

As in Khwaja and Mian (2008) we observe that many borrowers in our sample have multiple bank connections. Using the entire CIR database we select the 769,222 borrowers with loans from more than one bank. We do so for each quarter between 1988:II and 2006:IV. We are left with 13,265,830 firm-quarters. In each quarter we classify all borrowers into two groups, "risky" and "safe", according to their RECENT BAD CREDIT HISTORY (the dummy equals one for risky and zero for safe borrowers). Similarly in each quarter we classify all banks into small and large according to their TOTAL ASSETS (small banks are below and large banks are above the quarter median, respectively), as theory surmises small banks may be more affected by the stance of monetary policy in their ability and incentive to take on more credit risk.

For each borrower-quarter, we then calculate the difference between the percentages of the loan amounts that are obtained from either the borrower's small or large banks, a difference we surmise that will remain unaltered if a borrower's demand for financing changes.²⁸ We label this difference SMALL – LARGE BANK BORROWING. We first-difference this variable and regress it on the lagged INTEREST RATE, RECENT BAD CREDIT HISTORY, and their interaction. We further include level of borrower's debt, borrower unobservable effects and macro controls. We report the results of this regression in Table 3, Model I.

The negative and significant coefficient on INTEREST RATE suggests that following a monetary expansion small banks provide relatively more financing to *all* borrowers than large banks. More importantly, the negative and significant coefficient on the interaction term INTEREST RATE * RECENT BAD CREDIT HISTORY (i.e., -0.038**) suggests that following a monetary expansion small banks provide more loans to *risky* borrowers than large banks *vis-à-vis* loans to safer borrowers. The results suggest that, when interest rates are lower, large banks loosen their lending standards for risky firms less than small banks, suggesting that it is the *banks* that also *adjust* their credit risk-taking following changes in monetary policy (and not the risky borrowers that alter their demand).

Next we select only those borrowers that simultaneously have relationships with both small and large banks, leaving 6,844,387 firm-quarters. In consequence, in this

^{27.} Similarly Jiménez and Saurina (2006) find that lending standards worsen during good times. They consequently argue for countercyclical prudential norms.

^{28.} Following monetary changes, small banks are more affected in their risk-taking than large banks [see for example Stiglitz and Greenwald (2003) and Diamond and Rajan (2006)].

case borrowers have access to bank finance from both banks with high and low appetite for credit risk. Results in Model II (Table 3) strongly suggest that following lower interest rate small banks lend relatively more to borrowers with a bad credit history than large banks (the coefficient in this case equals -0.091***). We also replace the macro controls with 74 quarterly dummies. Results are unaffected and not reported.

4.3 Duration Analysis

While the results of the probit models and within borrower comparisons so far are suggestive *per se*, a number of additional steps are vital to reach firmer conclusions. First, banks may lend more to risky borrowers because they think that economic fundamentals are strong enough that the probability of default is reduced when rates are low (as low rates imply high borrowers' net worth for example). Hence, to analyze whether monetary policy influences credit-risk appetite we need to assess whether individual loans have a higher default probability or not.

Second, average loan maturity may change over time; worse, the stance of monetary policy may affect liquidity risk and hence loan maturity (more risk-taking would entail lending at longer maturities). Consequently, we need to have a measure of credit risk-taking that is normalized per period of time. Therefore, we need to analyze credit risk-taking employing duration models to assess how the hazard rate is affected by monetary policy.

4.4 Methodology

We analyze the time to default of the individual bank loan, or "loan spell", as a measure of risk.²⁹ We define default to occur if three months after the date of maturity or the date of an interest payment, the debt balance remains unpaid.³⁰ The hazard function in duration analysis provides us with a suitable method for summarizing the relationship between the time to default and the likelihood of default.³¹ The hazard rate effectively is a per-period measure of credit risk-taking and has an intuitive interpretation as the per-period probability of loan default provided the loan 'survives' up to that period. We rely on a parametric Weibull specification to determine the shape of the hazard function with respect to time. The Weibull distribution allows for monotonically positive or negative duration dependence, i.e., loans may be more or less likely to default as the time since their origination increases. As almost 80% of the loans in the sample are four quarters or shorter in maturity, we focus on the results for the standard Weibull parameterization with non-time varying covariates.³²

Repayment of a loan may prevent us from ever observing a default on this loan. Such a loan spell can be considered right censored. Not knowing when the default would occur means we are unable to observe the "true" time to default for these loan spells. With no adjustment to account for censoring, maximum likelihood estimation of the proportional hazard models produces biased and inconsistent estimates of model

^{29.} As in McDonald and Van de Gucht (1999) for example. Loans to small firms typically carry a relatively short maturity, often without early repayment possibilities; hence, we choose to ignore early repayment behavior captured in their competing risk model.

^{30.} We use "default" in the common sense of "a failure to pay financial debts" (*Merriam-Webster Dictionary*). Ninety days overdue is a standard period to classify a loan as non-performing but some countries use different overdue dates depending on the credit product [see for example Beattie, Casson, Dale, McKenzie, Sutcliffe and Turner (1995)].

^{31.} Cameron and Trivedi (2005), Heckman and Singer (1984), Kiefer (1988), Kalbfleisch and Prentice (2002) and Greene (2003) provide comprehensive treatments of duration analysis. Shumway (2001), Chava and Jarrow (2004) and Duffie, Saita and Wang (2007) discuss and employ empirical bankruptcy models.

^{32.} Our main results are unaffected either if we use a log-logistic specification which allows for a non-monotonic duration dependence or a Cox (1972) proportional hazard model for which the baseline hazard is a loan-specific constant. We find weak evidence for non-monotonicity at longer maturities. We return to estimation with time-varying covariates in the robustness subsection.

parameters. Accounting for right-censored observations will be accomplished in duration analysis by expressing the log-likelihood function as a weighted average of the sample density of completed loan spells and the survivor function of uncompleted spells. We return to the independence of the censoring scheme itself in the robustness subsection. As our sample consists out of *new* loans granted from 1988:II onwards, there is no left censoring problem.

4.5 First Results

The estimates in Table 4 are based on the maximum likelihood estimation of a duration model with a Weibull distribution. A positive coefficient indicates that an increase in the independent variable increases the hazard rate. All estimates are adjusted for right censoring and standard errors are clustered at the borrower level.

Model I controls for key macroeconomic conditions (specifically for the growth in real gross domestic product, the inflation rate, a measure of country risk, and two time trends, in level and squared), and also includes bank random effects to control for unobservable heterogeneity related to different bank lending behavior ("frailty model"). The estimated coefficient on INTEREST RATE_{τ -1} in Model I equals -0.099^{***} . Hence, a lower short-term interest rate prior to loan origination implies that banks grant loans with a *higher* hazard rate, i.e., with a higher probability of default per quarter. Therefore, we find support for H2. On the other hand, the coefficient on INTEREST RATE_{τ -1} equals 0.344^{***} indicating that a lower short-term rate during the life of the loan implies a *lower* loan hazard rate.

While lower short-term interest rates in the economy increase the hazard rate on new loans but decrease the hazard rate on the outstanding loans, higher GDP growth decreases the hazard rate on both new and outstanding loans (the coefficients on GDP growth are -0.201*** and -0.045***, respectively). These estimated coefficients also benchmark the effects of monetary policy on credit risk: A one percentage point change in the interest rate for example has similar sized effects as one percentage point change in GDP growth, though its standard deviation is almost double the size of the standard deviation of GDP growth one (more on economic relevancy later). All in all, these results suggest that expansive monetary policy, in contrast to economic growth, whets the appetite for credit risk.³³

Both higher inflation and country risk at origination entail as expected a higher loan hazard rate, while the positive coefficients on both time trends indicates a secular increase in Spanish bank risk-taking. The latter finding clearly demonstrates that more risk-taking need not be excessive *per se* and that only intertemporal variation in risk-taking across borrowers and banks can provide evidence in this regard.

Finally, the estimate of $\ln(\alpha)$, which is the logarithm of the parameter of duration dependence in the Weibull distribution, equals 0.620^{***} . As α equals 1.859 and is larger than one, the positive duration dependence we find implies that default becomes conditionally more likely over the life of the loan.

4.6 Bank, Borrower and Loan Characteristics

Bank characteristics have significantly changed during the entire sample period. We therefore add time variant bank characteristics in Model II. The key results we already discussed are basically unchanged. The coefficients on most bank characteristics are statistically significant,

^{33.} GDP growth also has an opposite effect than short-term rates on the selection of borrowers' quality of credit history (see Table 2).

except the coefficient on total assets. The coefficients on the own funds and liquidity ratio for example are negative, indicating that banks with more own funds at stake or with more liquidity grant loans with lower hazard rates. More own funds at stake may lower the incentive to take risk as in Keeley (1990) for example and, banks may hoard liquidity and therefore not grant risky loans as in Diamond and Rajan (2006). The coefficient on the bank's non-performing loan ratio (compared to the sector's average) is positive suggesting that banks seemingly persist in hazardous lending. Listed commercial banks, and more intensely savings banks and cooperatives grant more risky loans (0.154***; 0.422*** and 0.371***, respectively).

We further add borrower and loan characteristics in Model III of Table 4 to control for changes over time in the borrowers and loans and to disentangle supply from demand effects. A higher demand from risky borrowers or a deterioration in the quality of the marginal new borrower and/or project for example may lead banks with a constant risk-appetite to lower the amount of credit, to demand more collateral and/or to shorten maturity. Including these proxies for lending standards, and the borrower characteristics, leaves the main results basically unchanged.

With regard to the borrower characteristics, we observe that the coefficient on the proxy of the size of the firm is insignificant, whereas as the borrower ages the likelihood to default decreases. The risk profile of a firm, measured by its credit history, shows that firms with bad credit records in the past are more likely to default in the future. Finally, we observe that borrowers with a large number of bank relationships are more risky. Loan characteristics play also an important role explaining the time to default of a loan. Model III shows that smaller, more collateralized and short-term loans are riskier than the other [as in Jiménez, Salas and Saurina (2006)].

Controlling also for unobservable borrower heterogeneity employing random effects (corresponding to 39,963 individual firms) in Model IV, again does not alter our findings. All these results suggest that banks do *adjust* their credit risk-taking in response to changes in short-term rates. The coefficients on most borrower and loan characteristics are statistically significant in Model IV, except the coefficient on collateral.³⁴ Younger firms with more bank relationships are more likely to default and so are smaller loans with a shorter maturity. Once we control for unobservable firm heterogeneity the coefficient on the size of the borrower turns negative, showing that loans to smaller firms are riskier. Banks seem to adjust loan amount and maturity when the demand from risky borrowers is higher (again not including loan characteristics leaves the results unaffected).³⁵

Finally, in Model V we introduce interactions between the interest-rate-at-origination variable and a borrower bad credit history dummy and borrower age. Remember that one of our two main motivations to employ duration analysis was that more lending to borrowers with a bad or no credit history not necessarily implied more bank credit risk-taking if lower interest rates significantly bolstered borrowers' net worth for example. However, the estimates reported in Model V suggest that borrowers with a bad or shorter credit history fail even more when monetary policy is expansive. This is a key finding providing our

^{34.} The coefficient on collateral turns statistically significant when we do not control for unobservable borrower heterogeneity (as in Model III for example). Collateral may be set for the borrower in beginning of the relationship and may be only infrequently adjusted.

^{35.} For credit lines we take the total amount that is made available to the borrower. Dropping credit lines that are not drawn or dropping all credit lines does not alter the results. Some loans are also flagged as renewals. Dropping renewals does not change the results.

results on borrower selection in Tables 2 and 3 with a clearer interpretation: When interest rates are lower, there is more credit risk-taking. Therefore, we find support for H2.

4.7 Economic Relevancy and Dynamics

In this section we investigate how the stance and dynamic path of monetary policy affect credit risk. We employ the coefficients of the Model IV in Table 4 to calculate a quarterly probability of default for a loan with a mean maturity of five quarters,³⁶ and other mean characteristics, for different combinations of INTEREST RATE_{r-1} and INTEREST RATE_{r+T-1}. Figure 2 displays these combinations.

For example, if the short-term interest rate in the economy is equal to 4.1% (the sample mean) at loan origination and maturity, the quarterly probability of loan default is estimated to equal 0.6% (which equals the sample quarterly probability of loan default). In contrast, if the interest rate at origination equals 2.0% (the sample mean minus one standard deviation), but increases over the life of the loan to 6.2% (the sample mean plus one standard deviation), the default rate of the loan more than doubles to 1.4%. On the other hand, if the "path is reversed" and the funds rate drops from 6.2% to 2.0%, the default rate more than halves to 0.2%.

These results suggest that during long periods of low interest rates banks may take on more credit risk and relax lending standards. Exposing the "hazardous" cohort of loans, granted when rates were low, to swiftly increasing policy rates dramatically exacerbates their risk, these estimates suggest (an effect which is more muted when rates were high at origination).³⁷ Therefore, we also find support for H3.

To further assess the dynamics of credit risk-taking, we first-difference the interest rate and GDP growth variables and include additional lags of the interest rate in Model VI of Table 4. The effect of a change in the interest rate on the hazard rate is immediate. Coefficients on the quarterly change in the interest rate one and two quarters prior to origination are equal to -0.601^{***} and -0.152^{**} , respectively, while the coefficient on its third lag is small and not significant. The coefficient on the change in the interest rate over the life of a loan equals $+0.201^{***}$. These estimates again vividly illustrate the conundrum central banks may face when changing policy rates: On the one hand, lowering rates reduces on the short-run the credit risk of outstanding loans and increases the credit risk-taking of banks; on the other hand, there will be more defaults in the medium term.

Finally, we report the results for a duration model that allows the short-term interest rate, GDP growth rate, country risk and inflation to vary over the life of the loan (indexed $\tau + t - 1$, with $t: 1 \rightarrow T$). We focus on the time variation in only these variables as a number of bank and borrower characteristics do not vary much over time (bank ownership, number of borrower relationships, and bad credit history for example) or are deterministically "defined" with respect to the duration of the loan (age for example).

^{36.} The choice of maturity matters also because the estimated parameter of duration dependence is significantly larger than one: In Model IV it equals $e^{0.816} = 2.261$. The hazard rate will therefore increase over the life of the loan. Integrating the hazard rate over the life of the loan yields the probability of default of the loan.

^{37.} While suggestive of the impact of changes in monetary policy on the loan hazard rates, the estimates are calculated for one loan cohort only. To obtain a correct assessment of a monetary policy path on the aggregate hazard rate, cohort size and timing needs to be properly accounted for (loans granted during the period of the increase in the policy interest rate will have a lower and lower hazard rate for example). We leave such an exercise for future work.

In addition, the loan characteristics if allowed to vary may *not* be longer "ancillary" (or exogenous) with respect to loan duration!³⁸

Models VII and VIII report the estimates employing 1,989,170 loan-quarter observations. The coefficient on the interest rate prior to origination remains negative and significant, but the coefficient on the newly introduced time-varying interest rate covariate is small in Model VII. However, also the two trend variables strikingly loose significance in that model thus suggesting multicollinearity. Therefore, we drop the two time trend variables in Model VIII and the coefficient on the time-varying interest rate as expected returns to a value closer to previous specifications (i.e., non-time-varying).

Note also in Model VIII that when we allow inflation to vary over the life of the loan, higher inflation implies lower hazard rate. Higher inflation reduces the real value of the debt thus reducing default risk [see Allen and Gale (2004) and Diamond and Rajan (2006)]. However, higher inflation at origination implied more risk in Model I for instance.³⁹

4.8 Interactions with Bank Characteristics

Next we introduce interactions of the overnight rates with bank characteristics. Theory suggests banks' propensity to adjust credit risk-taking in response to monetary policy conditions may depend upon bank size, liquidity, type of ownership, and local market characteristics. Interactions can further assist us in addressing identification problems. We report the estimates in Table 5. To focus and conserve space, the coefficients on borrower and loan characteristics and the macro controls are no longer reported. These coefficients remain virtually unaffected from Table 4.

Smaller banks not only take more credit risk but also change their credit risk-taking more in response to changes in the stance of monetary policy (Model I in Table 5). Their lower net worth, lack of diversification, and more difficulties in accessing liquidity may provide an explanation [Diamond and Rajan (2006)]. On the other hand, banks with more liquidity take more risks and respond more to a lowering of interest rates in their risk-taking than other banks (Model II). Banks may hoard liquidity (possibly because they take higher risk). Hence when short-term interest rates are lower, banks with high levels of liquidity may finance more risky long-term projects [Diamond and Rajan (2006)], as they no longer need this level of liquidity since their cost / access to liquidity improves. Liquid bank assets may also worsen the moral hazard problem as these assets can be easily directed towards risky ventures [Myers and Rajan (1998)]. When monetary policy is expansive and bank return on assets is low, the banks that can increase credit risk-taking more are those with many liquid assets at hand.⁴⁰

^{38.} To obtain interpretable estimates it is required that the variables be either "defined" or "ancillary" with respect to the duration of the loan. A defined variable follows a deterministic path. *Age* is an example of a defined variable because its path is set in advance of the loan and varies deterministically with loan duration. An ancillary variable has a stochastic path but the path cannot be influenced by the duration of the loan. *Collateralization* for example is probably not ancillary as banks may eventually tighten collateral requirements when hazard rates increase.

^{39.} Higher inflation may amplify the standard deviation of the spread between bank loan and deposit rates, yet cut bank profitability and lead to banking instability (Boyd and Champ (2003)]. The evidence seems sometimes mixed though. In Beck, Demirgüç-Kunt and Levine (2003) for example lower inflation eases financing obstacles faced by borrowing firms, while in Demirgüç-Kunt and Detragiache (2002) inflation does not help predict the occurrence of banking crises [in contrast to Demirgüç-Kunt and Detragiache (1998)].

^{40.} The measure for bank liquidity we employ is often used in the literature that investigates the credit channel of monetary policy [see Kashyap and Stein (2000) for example]. In a previous version of this paper and in non-reported regressions, we explore other measures of liquidity such as the ratios of loans to deposits and interbank deposits

Banks with a relatively higher ratio of non-performing loans seem to continue in their ways by granting loans with a higher hazard rate (Model III). However monetary policy does not affect their credit risk-taking in a statistically significant way.⁴¹ While credit cooperatives grant loans with a higher hazard rate than private commercial banks, their risk-taking is not significantly affected by monetary policy (Model IV). Savings banks increase less their risk-taking when rates are lower, which suggests that bank managers seem to be more cautious. That might be the result of a more limited upside potential in their pay packages since savings banks have no shares. Alternatively, they might try to rein more on credit risk since their portfolios are, *ceteris paribus*, always more risky (the SAVINGS BANK parameter is always positive, significant and the largest one among type of ownership dummy variables in specifications without interaction terms).

Having more new borrowers in the province increases the hazard rate on new loans (Model V), as the adverse selection problem among banks stemming from bank relationships with old borrowers is alleviated thereby weakening the bank incentives to screen. When monetary policy is more expansionary, the relative cost of screening versus non-screening increases, providing incentives to banks to soften their lending standards [Dell'Ariccia and Marquez (2006)]. We find that the presence of more new borrowers in the province reinforces the impact of short-term rates on credit risk-taking.

Finally, with more banking competition (proxied by a lower Herfindahl-Hirschman Index), banks have more incentives to take risk because the franchise value of the bank is lower [as in Keeley (1990)]. Thus, with easy access to liquidity during monetary expansions, a very competitive environment for banks may enhance risk taking [Dell'Ariccia and Márquez (2006), Sengupta (2007)].

Overall, we find that the impact of the stance of monetary policy on credit risk-taking depends on bank size, liquidity, and ownership. Local banking conditions, at the province level, like the percentage of new borrowers and the banking market concentration also matter. Therefore, the evidence is consistent with H4.

4.9 Robustness

Finally, in Table 6 we run through a number of additional robustness checks. To conserve space we now also suppress the coefficients on the bank characteristics. These coefficients remain virtually unaffected from Table 5.

4.9.1 CENSORING SCHEME INDEPENDENCE: LOAN MATURITY AND MULTIPLICITY

Banks may not only change loan risk, but also alter other loan terms according to monetary conditions. For instance, expansive monetary policy could make banks more willing to bear liquidity risk [Diamond and Rajan (2006)]. Changes in observed loan maturity in particular are a fundamental reason to rely on duration and not on probit models when assessing loan risk on the basis of observed loan default. But even if the loan maturity (at origination) would be directly affected by monetary policy (the conditional correlation coefficient of –0.08 actually suggests it is *not*),⁴² or borrowers would repay early to benefit from a lower interest

⁽a measure for the importance of wholesale depositors). We find that the impact of interest rates on credit risk-taking also depends on the structure of the deposits.

^{41.} The insignificant coefficient on the interaction term is positive, possibly because banking supervisors face a capacity constraint (when low interest rates spur risk-taking by many banks) and focus on curtailing risk-taking by banks with high NPL ratios.

^{42.} The correlation is even weaker when we consider the Euro period (i.e., it equals 0.01).

rate on a new loan, the censoring scheme would remain independent and the basic methods of survival analysis would remain valid.⁴³

However, schemes in which the failure times of loans are censored because of an unusually high (or low) risk of failure are not independent and the basic methods of survival analysis are then not valid. Early repayments of individual loans for example that would follow from their lower failure rates do not constitute an independent censoring mechanism. However this problem should be limited as most loans have a short maturity and do not have early repayment clauses, nor adjustable rates.

Nevertheless, we conservatively remove all loans with maturity longer than one year in Model I in Table 6. Despite losing more than 20% of all observations (with 536,571 observations remaining), results are mostly unaffected. For selected specifications we also check the results for loans with maturity shorter than three months. Again results are unaltered. Note again that these loans with short maturities do not have adjustable interest rates. Therefore, our findings related to the effect of overnight rates on credit risk seem not driven by the higher direct financing costs resulting from the adjustable loan rates. We also remove all loans with maturity shorter than one year. Results (unreported) do not change qualitatively despite working again with a much smaller number of loans (137,556 observations). In sum, overnight rates influence the hazard rate across the maturity spectrum.

Repayments by the same borrower may also not be independent *across* loans. If successful in servicing other loans, borrowers may strive to avoid defaulting on the "last" loan for example. Alternatively, a default on one loan may increase the probability of default on the other loans. Our earlier borrower random effects model and clustering at the borrower level throughout all other exercises should have accounted for this dependence. Nevertheless, next we also remove all the loans from borrowers that have multiple loans outstanding. Only 234,338 "single" loans remain. This rather drastic loan selection, however, does not alter the results (Model II of Table 6). Hence, H2 seems again supported, irrespective of the maturity and multiplicity of the loans.

4.9.2 EXOGENEITY OF MONETARY POLICY

Model III in Table 6 focuses on the 1988-1998 period during which the Spanish peseta was pegged to the German mark. In this earlier period most sample borrowers also had no access to bond financing while securitization of loans was almost inexistent. In fact, if we consider all the loans that were securitized (and most of them were mortgages, not business loans), securitized loans represented less than 1% of the total number of loans in 1998 (but already 15.3% of all loans in 2006). In consequence, borrowers in Spain in this earlier period were almost entirely dependent on bank credit and banks did not transfer the credit risk. While the number of loan observations drops to 292,346 once we consider the 1988-1898 period, the results are mostly unaffected however.

Next, we replace again both interest rate variables by the interest rate in Spain, instrumented by the German interest rate, but do not detrend. As in Table 2, in a first stage we regress the INTEREST RATE in SPAIN on all predetermined variables including

^{43.} A censoring scheme is said to be independent if the probability of censoring at each time *t* depends only on random processes that are independent of the failure times in the trial, the observed pattern of failures and censoring up to time *t* in the trial, or (as in our case) on a covariate [Kalbfleisch and Prentice (2002), p.13].

the German overnight interest rates (see Table 2B). The estimated coefficients on the instrumented INTEREST RATE SPAIN are similar to the estimates in the duration models using the INTEREST RATE variables.⁴⁴ All in all, Model IV in Table 6 shows that using the (instrumented) Spanish interest rate does not alter the results.

In Model V we report the equivalent results for the 1999-2006 Euro period (381,781 loans). Results for both periods are strikingly similar. It is interesting to notice that nominal rates were in some sub-periods very low compared to a Taylor rule, since in these sub-periods the three largest Euro area economies had significant lower levels of GDP growth and inflation than Spain. In addition, this period of time is more homogenous: Inside this period the Spanish economy in general, and the banking system in particular, did not experience any significant structural change. Therefore, H2 seems supported either for the whole period or for each of the two subperiods separated by the inception of the Euro, which further helps in giving the results a causal interpretation.

4.9.3 DEFINITIONS OF INDEPENDENT VARIABLES AND OMITTED VARIABLES

The time trends we included so far may capture improvements in the efficiency in the Spanish banking sector during the last 20 years. The efficiency of the banking sector has dramatically improved in Spain, potentially biasing our results if we would not control for this effect. Consequently, we also replace the trend variables once with the variables EFFICIENCY RATIO and FINANCIAL INCOME/ATA, respectively. The EFFICIENCY RATIO, a direct measure of efficiency, is defined as expenses over gross operating margin of all banks. FINANCIAL INCOME/ATA equals the interest income plus dividends received over average total assets of all banks (i.e., asset profitability), thus, reflecting a more indirect measure of efficiency. Alternatively, it controls for the risk profile of the bank (on average, riskier banks should have a higher FINANCIAL INCOME/ATA ratio). Results are virtually unaffected and we choose not to report them. Time trends may also capture the volatility of GDP growth. Replacing the time trends with these variables again does not affect our findings.

Changes in credit composition [with respect to risk for example) not necessarily affect the volume of credit (Matsuyama (2007)]. However, in the banking lending channel, monetary policy stance affects the volume of lending, maybe implying more credit-risk if the marginal borrowers were of a lower quality. In Table 6, Model VI we control for the annual growth rate in business loans for each individual bank at the time of the origination of the loans.⁴⁵ We find that results are very similar suggesting that the stance of monetary policy may imply a change in credit risk-taking and an optimal change in credit composition in addition to changing the credit volume (which has a positive and significant parameter, as expected).⁴⁶

Note that this regression is also related to the demand *versus* supply identification issue. If banks would want to keep their level of credit risk constant, then more credit volume

^{44.} The required adjustment of the standard errors on the estimated coefficients is not immediately available. The difference in fit of the models estimated with the actual and the projected Spanish interest rate, however, suggests that the adjustment factor is likely to be close to one. We further calculate the adjustment factor for a model estimated using ordinary least squares for which the dependent variable equals the logarithm of the time to default (censored observations are set equal to their maximum, i.e., 48 months). Again the adjustment factor is close to one and the estimated coefficients themselves in this linear model are very similar to those of the duration model.

^{45.} For comparability reasons we use the annual growth rate. Using the more noisy quarterly growth leaves results virtually unaffected.

^{46.} As in Jiménez and Saurina (2006) who find that "excess" credit growth has a positive impact on future ex post credit losses.

would imply (if the marginal projects were riskier) a tightening of lending standards or maybe a different pool of borrowers. In Tables 4 to 6 we control for loan characteristics such as collateral, maturity and loan amount, and we also control for some borrower characteristics including identity. Hence, our results suggest that the stance of monetary policy in general and in particular the level of overnight rates also influence the appetite of banks for credit risk.

Lower interest rates may increase real estate prices and the value of collateral. This may imply more lending as in Kyotaki and Moore (1997), but defaults may occur once liquidity evaporates.⁴⁷ Model VII controls for the changes in house prices. The results suggest the other mechanisms we identified so far are at work besides this channel.⁴⁸

Monetary policy may further react to lower expected GDP growth, even if determined by central banks in which price stability is the main objective of monetary policy. Lower forecasted GDP growth (that is, potentially higher risk in the future) would result in lower central bank policy rates in this case – a reverse causality problem. We introduce expected GDP growth in Model VIII to further deal with this problem. We focus on the Euro period for two reasons: (1) we have access to the ECB's own quarterly predictions and the predictions by the Organization for Economic Co-operation and Development (OECD) for the entire period only come at a semiannual frequency; and (2) the business cycle conditions in the Euro area are more relevant (than the German conditions in the earlier period) as the now opened Spanish economy is more connected to the larger Euro area. In addition, by including expected GDP growth in the Euro Area, we control as well for the possibility that the results were driven by changes in Euro Area GDP growth rather than by ECB rates. However we find very similar results.⁴⁹

Finally, in Model IX we introduce Spanish long-term interest rates prior to loan origination, in particular the interest rate on the SPANISH TEN YEAR BOND, which is the most liquid one. Long-term rates may also matter for credit risk-taking. However, results are again unaffected, maybe because banks finance themselves more through short maturity debt that is more affected by short-term rates.

^{47.} While we cannot control for loss given default, empirical evidence actually shows a negative correlation between default probabilities and average recovery rates [Altman, Brady, Resti and Sironi (2005) and Acharya, Bharath and Srinivasan (2007)]. In addition to house prices, industry, province and borrower effects may have absorbed differences in recovery rates.

^{48.} We control at the firm level for the level of debt. This is important as lower interest rates make debt cheaper, which may increase debt levels and make firm defaults more likely. However, comparing the results in Models I and II in Table 4 (no debt level variable) with all subsequent models suggest that the level of debt is not the only channel through which the stance of monetary policy influences the hazard rate.

^{49.} In a non-reported regression we also use OECD forecasts for the whole period and control for German GDP growth. Results are virtually the same.

5 Conclusions and Policy Implications

This paper is the first to empirically investigate four questions that tie overnight rates and the stance of monetary policy to credit risk-taking and that are relevant for both academics and policy makers (central bankers and supervisors alike). Do lower short-term interest rates lead to more lending to borrowers that in the past were riskier? Do lower interest rates increase the credit risk-taking by banks? Are default rates the highest when after a period of low interest rates, which boosted credit risk-taking, a significant monetary tightening devalues the net worth of the borrowers? Do the answers to the former questions vary across types of banks, borrowers and banking markets?

We employ the Spanish Credit Register, a unique and comprehensive dataset that contains almost all business loan contracts from the last twenty-three years, and rely on the short-term interest rates set by the *Bundesbank* and then the ECB as the proper, yet fairly exogenous, measure of the stance of monetary policy. Therefore we can identify the effect of short-term interest rates on credit risk-taking. We are able to tackle the pernicious endogeneity problem that arises if monetary policy is determined domestically, disentangle demand of risky credit from supply and control for alternative hypotheses.

Controlling for macroeconomic conditions and for bank, loan, and borrower characteristics, we find robust evidence that suggests that prior to loan origination lower short-term interest rates motivate banks to soften their lending standards and grant more loans to borrowers with a bad or no credit history. More importantly, banks grant loans with a higher hazard rate (default probability normalized per time, a desirable normalization since monetary policy may also affect liquidity risks and hence maturity). In consequence, these results suggest that following a monetary expansion better borrowers' net worth and a higher appetite for liquidity risk are not the only motives for the banks' new engagements [see Bernanke, Gertler and Gilchrist (1996), Diamond and Rajan (2006) and Matsuyama (2007)]. Following a monetary expansion banks want to take more credit risk as well [see Smith (2002), Stiglitz and Greenwald (2003), Diamond and Rajan (2006), Dell'Ariccia and Marquez (2006), Rajan (2006), and Borio and Zhu (2007)].

Conditioning on the loan being granted, lower interest rates imply lower credit risk —i.e., lower interest rates reduce the credit risk of outstanding loans. This is possibly the case because refinancing costs are lower and borrower net worth is higher and, therefore, credit risk is lower. Consequently there is a completely different impact of lower interest rates on the credit risk of new *vis-à-vis* outstanding loans.

In sum, in the short-run lower interest rates reduce total credit risk of banks since the volume of outstanding loans is larger than the volume of new loans. In the medium term, lower interest rates, however, may increase credit risk in the economy. In particular, a period of low interest rates followed by a severe monetary contraction maximizes credit risk, as the already "hazardous" cohort of new loans gets exposed to higher interest rates as outstanding loans. On the other hand, steep declines in interest rates minimize total credit risk thus possibly reducing a credit crunch.

The impact of monetary policy on risk-taking is not equal for all banks: Small banks, banks that are flush with liquidity and commercial banks take on more extra risk when interest

rates are low. Therefore, balance-sheet strength, investment opportunities, moral hazard and type of bank ownership shape the impact of monetary policy on bank credit risk-taking. In addition, more new firms and fiercer competition in the market enhance the impact of interest rates on bank credit risk-taking [see Dell'Ariccia and Marquez (2006) and Keeley (1990)].

We also find that higher GDP growth reduces credit risk for new and outstanding loans alike and for loans to borrowers with bad credit history. Higher GDP growth or lower short-term interest rates imply higher borrower net worth and, therefore, fewer agency problems between lenders and borrowers. However, the effect of GDP growth on credit risk-taking is different from the effect of short-term interest rates. This implies that there may be other financial inefficiencies [Rajan (2006) and Diamond and Rajan (2006)] that explain the results of this paper. Finally, higher inflation during the life of the loan reduces credit risk, maybe because it reduces the real value of debt [Allen and Gale (2004) and Diamond and Rajan (2006)]. In contrast higher inflation before loan origination implies higher risk taking. Finally, we find that long-term interest rates have a weaker effect on credit risk-taking than short-term rates. Short-term rates possibly affect bank risk-taking more because banks finance themselves mainly through short-term debt [Diamond and Rajan (2006)].

There are a number of natural extensions to our study. First, we currently focus on the impact of monetary policy on the hazard rate of individual bank loans but overlook the correlations between loan default and the impact on each individual bank's portfolio or the correlations between all the banks' portfolios and the resulting systemic impact of monetary policy. Second, given the cohorts of loans and initial and ending policy rates for a time period, one can calculate on the basis of the estimated coefficients the path of monetary policy rates that would minimize the total amount of credit risk. It would be interesting to compare this path to the actual path that was followed. Third, we have studied the effects of monetary policy on the composition of credit in only one dimension, i.e., risk. Industry affiliation or portfolio distribution between mortgages, consumer loans and commercial and industrial loans for example may also change. Fourth, we focus on one of the possible causes of the current credit crisis, but we were silent on whether the risk-taking we observe was excessive. In credit channel models banks are usually too conservative due to agency problems with their borrowers and expansive monetary policy can therefore not result in excessive risk-taking. However, in Rajan (2006) or in banking models with agency problems between shareholders and debtholders and/or banking regulators [Freixas and Rochet (2008)], banks will take excessive risk. The structure of bank deposits, debt, ownership and control ----in particular, for listed banks, board independence and ownership dispersion and, for non-listed commercial banks, savings banks and credit cooperatives, stakeholders' relationships-may therefore determine the impact of monetary policy on credit risk-taking. Finally, the impact of monetary policy on credit risk-taking by banks may be amplified by certain types of financial innovation, for example loan securitization. Given space constraints we leave these extensions for future work.

REFERENCES

ACHARYA, V. V., S. T. BHARATH and A. SRINIVASAN (2007). "Does Industry-wide Distress Affect Defaulted Firms? Evidence from Creditor Recoveries", *Journal of Financial Economics*, 85, pp. 787-821.

ADALID, R., and C. DETKEN (2007). Liquidity Shocks and Asset Price Boom/Bust Cycles, European Central Bank, Frankfur.

AHREND, R., B. COURNÈDE and R. PRICE (2008). Monetary Policy, Market Excesses and Financial Turmoil, Organisation for Economic Co-operation and Development, Paris.

ALLEN, F., and D. GALE (2004). "Asset Price Bubbles and Monetary Policy", in M. Desai and Y. Said (Eds.), *Global Governance and Financial Crises*, Routledge, London, pp. 19-42.

ALTMAN, E. I., B. BRADY, A. RESTI and A. SIRONI (2005). "The Link between Default and Recovery Rates: Theory, Empirical Evidence, and Implications", *Journal of Business*, 782203-2227.

AXELSON, U., T. JENKINSON, P. STRÖMBERG and M. S. WEISBACH (2007). Leverage and Pricing in Buyouts: An Empirical Analysis, Stockholm School of Economics, Stockholm.

BANCO DE ESPAÑA (1997). La Política Monetaria y la Inflación en España (Monetary Policy and Inflation in Spain), Alianza Editorial, Madrid.

BEATTIE, V. A., P. D. CASSON, R. S. DALE, G. W. MCKENZIE, C. M. S. SUTCLIFFE and M. J. TURNER (1995). Banks and Bad Debts: Accounting for International Losses in International Banking, John Wiley & Sons, London.

BECK, TH., A. DEMIRGÜÇ-KUNT and R, LEVINE (2003). Bank Supervision and Corporate Finance, Cambridge MA: National Bureau of Economic Research.

BERGER, A. N., and G. F. UDELL (1995). "Relationship Lending and Lines of Credit in Small Firm Finance", *Journal of Business*, 68, pp. 351-381.

- BERNANKE, B. S. (2007). The Financial Accelerator and the Credit Channel, Board of Governors of the US Federal Reserve System, Washington DC.
- BERNANKE, B. S., and A. S. BLINDER (1988). "Money, Credit and Aggregate Demand", *American Economic Review*, 82, pp. 435-439.

---- (1992). "The Federal Funds Rate and the Channels of Monetary Transmission", American Economic Review, 82, pp. 901-921.

BERNANKE, B. S., and M. GERTLER (1989). "Agency Costs, Net Worth, and Business Fluctuations", *American Economic Review*, 79, pp. 14-31.

- (1995). "Inside the Black Box: The Credit Channel of Monetary Policy Transmission", Journal of Economic Perspectives, 9, pp. 27-48.
- BERNANKE, B. S., M. GERTLER and S. GILCHRIST (1996). "The Financial Accelerator and the Flight to Quality", *Review* of *Economics and Statistics*, 78, pp. 1-15.

— (1999). "The Financial Accelerator in a Quantitative Business Cycle Framework", in J. Taylor and M. Woodford (Eds.), Handbook of Macroeconomics, Elsevier, Amsterdam, pp. 1341-1393.

BERNANKE, B. S., and K. N. KUTTNER (2005). "What Explains the Stock Market's Reaction to Federal Reserve Policy?", *Journal of Finance*, 60, pp. 1221-1258.

BHARATH, S., S. DAHIYA, A. SAUNDERS and A. SRINIVASAN (2006). "So What Do I Get? The Bank's View of Lending Relationships", *Journal of Financial Economics,* forthcoming.

BHATTACHARYA, S. (1982). "Aspects of Monetary and Banking Theory and Moral Hazard", *Journal of Finance*, 37, pp. 371-384.

BHATTACHARYA, S., and G. CHIESA (1995). "Proprietary Information, Financial Intermediation, and Research Incentives", *Journal of Financial Intermediation*, 4, pp. 328-357.

BLACK, L. K., and R. J. ROSEN (2008). The Effect of Monetary Policy on the Availability of Credit: How the Credit Channel Works, Board of Governors of the Federal Reserve System, Washington DC.

BLANCHARD, O. (2006). Macroeconomics, Prentice Hall, New York.

BOIVIN, J., B. MOJON and M. P. GIANNONI (2008). "How Has the Euro Changed the Monetary Transmission?", in D. Acemoglu, K. Rogoff and M. Woodford (Eds.), NBER Macroeconomics Annual, MIT Press, Cambridge MA, forthcoming.

BORIO, C. (2003). "Towards a Macroprudential Framework for Financial Supervision and Regulation", CESifo Economic Studies, 49, pp. 181-215.

BORIO, C., and PH. LOWE (2002). Asset Prices, Financial and Monetary Stability: Exploring The Nexus, Bank of International Settlements, Basel.

BORIO, C., and H. ZHU (2007). Capital Regulation, Risk-Taking and Monetary Policy: A Missing Link in the Transmission Mechanism, Bank for International Settlements, Basle.

BOYD, J. H., and B. CHAMP (2003). Inflation and Financial Market Performance: What Have We Learned in the Last Ten Years?, Federal Reserve Bank of Cleveland, Cleveland.

CABALLERO, R. (2006). "On the Macroeconomics of Asset Shortages", in A. Beyer and L. Reichlin (Eds.), *The Role of Money: Money and Monetary Policy in the Twenty-First Century,* European Central Bank, Frankfurt, pp. 272-283.

CALOMIRIS, CH. W., and TH. PORNROJNANGKOOL (2006). *Relationship Banking and the Pricing of Financial Services,* Cambridge MA: National Bureau for Economic Research.

CAMERON, A. C., and P. K. TRIVEDI (2005). *Microeconometrics: Methods and Applications,* Cambridge University Press, New York.

CHAVA, S., and R. JARROW (2004). "Bankruptcy Prediction with Industry Effects", *Review of Finance*, 8, pp. 537-569. CHRISTIANO, L. J., M. EICHENBAUM and CH. EVANS (1996). "The Effects of Monetary Policy Shocks: Evidence from

- CORDELLA, T., and E. LEVY-YEYATI (2003). "Bank Bailouts: Moral Hazard versus Value Effect", Journal of Financial Intermediation, 12, pp. 300-330.
- COX, D., (1972). "Regression Models and Life Tables", Journal of the Royal Statistical Society, 24, pp. 187-201.

DEGRYSE, H., and S. ONGENA (2001). "Bank Relationships and Firm Performance", *Financial Management*, 30, pp. 9-34.

- DELGADO, J., V. SALAS and J. SAURINA (2007). "Joint Size and Ownership Specialization in Bank Lending", *Journal on Banking and Finance*, 31, pp. 3563-3583.
- DELL'ARICCIA, G., D. IGAN and L. LAEVEN (2008). Credit Booms and Lending Standards: Evidence from the Subprime Mortgage Market, International Monetary Fund, Washington DC.
- DELL'ARICCIA, G., and R. MÁRQUEZ (2006). "Lending Booms and Lending Standards", Journal of Finance, 61, pp. 2511-2546.
- DEMIRGÜÇ-KUNT, A., and E. DETRAGIACHE (1998). *The Determinants of Banking Crises in Developing and Developed Countries*, International Monetary Fund, Washington DC.
- (2002). "Does Deposit Insurance Increase Banking System Stability: an Empirical Investigation", Journal of Monetary Economics, 49, pp. 1373-1406.
- DEN HAAN, W. J., S. SUMNER and G. YAMASHIRO (2007). "Bank Loan Portfolios and the Monetary Transmission Mechanism", *Journal of Monetary Economics*, forthcoming.
- DIAMOND, D. W. (1984). "Financial Intermediation and Delegated Monitoring", *Review of Economic Studies*, 51, pp. 393-414.
- DIAMOND, D. W., and R. G. RAJAN (2006). "Money in a Theory of Banking", American Economic Review, 96, pp. 30-53.
- DUFFIE, D., L. SAITA and K. WANG (2007). "Multi-Period Corporate Default Prediction with Stochastic Covariates", *Journal of Financial Economics*, forthcoming.
- FARINHA, L. A., and J. A. C. SANTOS (2002). "Switching from Single to Multiple Bank Lending Relationships: Determinants and Implications", *Journal of Financial Intermediation*, 11, pp. 124-151.
- FOK, R. C. W., Y. C. CHANG and W. T. LEE (2004). "Bank Relationships and Their Effects on Firm Performance around the Asian Financial Crisis: Evidence from Taiwan", *Financial Management*, 33, pp. 89-112.

FREIXAS, X., and J. CH. ROCHET (2008). Microeconomics of Banking, MIT Press, Cambridge MA.

- GERTLER, M., and S. GILCHRIST (1993). "The Role of Credit Market Imperfections in the Monetary Transmission Mechanism: Arguments and Evidence", *Scandinavian Journal of Economics*, 95, pp. 43-64.
- (1994). "Monetary Policy, Business Cycles, and the Behavior of Small Manufacturing Firms", Quarterly Journal of Economics, 109, pp. 309-340.
- GREENE, W. H. (2003). Econometric Analysis, Prentice Hall, Upper Saddle River NJ.
- GRUNERT, J., L. NORDEN and M. WEBER (2005). "The Role of Non-Financial Factors in Internal Credit Ratings", Journal of Banking and Finance, 29, pp. 509-531.
- HECKMAN, J. J., and B. SINGER (1984). "Econometric Duration Analysis", Journal of Econometrics, 24, pp. 63-132.
- HELLMAN, T., K. MURDOCK and J. E. STIGLITZ (2000). "Liberalization, Moral Hazard in Banking and Prudential Regulation: Are Capital Controls Enough?", *American Economic Review*, 90, pp. 147-165.
- HERNANDO, I., and J. MARTÍNEZ-PAGÉS (2001). Is There a Bank Lending Channel of Monetary Policy in Spain?, European Central Bank, Frankfurt.
- HERTZBERG, A., J. M. LIBERTI and D. PARAVISINI (2007). Information and Incentives Inside the Firm: Evidence from Loan Officer Rotation, Kellogg School of Management Northwestern University, Chicago IL.
- IOANNIDOU, V. P., and S. ONGENA (2008). "Time for a Change": Loan Conditions and Bank Behavior When Firms Switch, CentER - Tilburg University, Tilburg.
- IOANNIDOU, V. P., S. ONGENA and J. L. PEYDRÓ (2007). Monetary Policy and Subprime Lending: "A Tall Tale of Low Federal Funds Rates, Hazardous Loans, and Reduced Loan Spreads", CentER - Tilburg University / European Central Bank, Tilburg.
- JIMÉNEZ, G., V. SALAS and J. SAURINA (2006). "Determinants of Collateral", Journal of Financial Economics, 81, pp. 255-281.
- JIMÉNEZ, G., and J. SAURINA (2004). "Collateral, Type of Lender and Relationship Banking as Determinants of Credit Risk", *Journal of Banking and Finance*, 28, pp. 2191-2212.
- (2006). "Credit Cycles, Credit Risk, and Prudential Regulation", International Journal of Central Banking, 2, pp. 65-98. KALBFLEISCH, J. D., and R. L. PRENTICE (2002). The Statistical Analysis of Failure Time Data, Wiley-Interscience, New York NY.
- KANE, E. J. (1989). The S&L Insurance Mess: How Did it Happen?, MIT Press, Cambridge MA.
- KASHYAP, A. K., and J. C. STEIN (2000). "What Do A Million Observations on Banks Say About the Transmission of Monetary Policy?", American Economic Review, 90, pp. 407-428.
- KEELEY, M. C. (1990). "Deposit Insurance Risk and Market Power in Banking", American Economic Review, 80, pp. 1183-1200.
- KEYS, B., T. MUKHERJEE, A. SERU and V. VIG (2008). Did Securitization Lead to Lax Screening? Evidence From Subprime Loans 2001-2006, Chicago Graduate School of Business, Chicago IL.
- KHWAJA, A., and A. MIAN (2008). "Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market", American Economic Review, forthcoming.
- KIEFER, N. M. (1988). "Economic Duration Data and Hazard Functions", Journal of Economic Literature, 26, pp. 646-679.

KRUGMAN, P. R., and M. OBSTFELD (2006). International Economics: Theory and Policy, Addison-Wesley, New York.

- KURITZKES, A., and T. SCHUERMANN (2008). "What We Know, Don't Know, and Can't Know about Bank Risk: A View from the Trenches", in F. X. Diebold, N. Doherty and R. J. Herring (Eds.), *The Known, The Unknown and The Unknowable in Financial Risk Management, Princeton; University Press, forthcoming.*
- KYOTAKI, N., and J. MOORE (1997). "Credit Cycles", Journal of Political Economy, 105, pp. 211-248.

LANG, W. W., and L. I. NAKAMURA (1995). "'Flight to Quality' in Banking and Economic Activity", *Journal of Monetary Economics*, 36, pp. 145-164.

LIBERTI, J. M. (2004). Initiative, Incentives and Soft Information: How Does Delegation Impact the Role of Bank Relationship Managers?, Kellogg School of Management Northwestern, Chicago IL.

MCDONALD, C. G., and L. M. VAN DE GUCHT (1999). "High-Yield Bond Default and Call Risks", *Review of Economics and Statistics*, 81, pp. 409-419.

MANGANELLI, S., and G. WOLSWIJK (2007). Market Discipline, Financial Integration and Fiscal Rules: What Drives Spreads in the Euro Area Government Bond Market?, European Central Bank, Frankfurt.

MARTIN-OLIVER, A., V. SALAS and J. SAURINA (2006). "Risk Premium and Market Power in Credit Markets", *Economics Letters*, 93, pp. 450-456.

MATSUYAMA, K. (2007). "Credit Traps and Credit Cycles", American Economic Review, 97, pp. 503-516.

MATUTES, C., and X. VIVES (2000). "Imperfect Competition, Risk Taking, and Regulation in Banking - An Incentive Structure for a Financial Intermediary", *European Economic Review*, 34, pp. 1-34.

MIAN, A., and A. SUFI (2008). The Consequences of Mortgage Credit Expansion: Evidence from the 2007 Mortgage Default Crisis, Cambridge MA: National Bureau for Economic Research.

MYERS, S., and R. G. RAJAN (1998). "The Paradox of Liquidity", Quarterly Journal of Economics, 113, pp. 733-771.

- ONGENA, S., G. TÜMER-ALKAN and N. VON WESTERNHAGEN (2007). Creditor Concentration: An Empirical Investigation, CentER Tilburg University, Tilburg.
- PETERSEN, M. A., and R. G. RAJAN (1994). "The Benefits of Lending Relationships: Evidence from Small Business Data", *Journal of Finance*, 49, pp. 3-37.

RAJAN, R. G. (2006). "Has Finance Made the World Riskier?", European Financial Management, 12, pp. 499-533.

RIGOBON, R., and B. Sack (2004). "The Impact of Monetary policy on Asset Prices", *Journal of Monetary Economics*, 51, pp. 1553-1575.

RUCKES, M. (2004). "Bank Competition and Credit Standards", Review of Financial Studies, 17, pp. 1073-1102.

SENGUPTA, R. (2007). Lending to Uncreditworthy Borrowers, Federal Reserve Bank of St. Louis, St. Louis MO.

SHILLER, R. J. (2007). Low Interest Rates and High Asset Prices: An Interpretation in Terms of Changing Popular Economic Models, Cowles Foundation for Research in Economics, New Haven CT.

SHUMWAY, T. (2001). "Forecasting Bankruptcy More Accurately: A Simple Hazard Model", Journal of Business, 74, pp. 101-124.

SMITH, B. D. (2002). "Monetary Policy, Banking Crises, and the Friedman Rule", *American Economic Review*, 92, pp. 128-134.

STAIGER, D., and J. H. STOCK (1997). "Instrumental Variables Regression with Weak Instruments", *Econometrica*, 65, pp. 557-586.

STIGLITZ, J. E. (2001). Information and the Change in the Paradigm in Economics, Nobel Foundation, Stockholm.

- STIGLITZ, J. E., and B. GREENWALD (2003). *Towards a New Paradigm in Monetary Economics,* Cambridge University Press.
- STIGLITZ, J. E., and A. WEISS (1981). "Credit Rationing in Markets with Imperfect Information", *American Economic Review*, 71, pp. 393-410.

TAYLOR, J. (2007). Housing and Monetary Policy, Federal Reserve Bank of Kansas City, Jackson Hole WY.

- TREACY, W. F., and M. S. CAREY (2000). "Credit Risk Rating Systems at Large U.S. Banks", *Journal of Banking and Finance*, 24, pp. 167-201.
- YOSHA, O. (1995). "Information Disclosure Costs and the Choice of Financing Source", Journal of Financial Intermediation, 4, pp. 3-20.

FIGURE 1. EMPIRICAL STRATEGY

The figure summarizes the empirical strategy we follow.



FIGURE 2. MONETARY POLICY PATHS AND LOAN HAZARD RATE

The figure displays on the vertical axis the quarterly probability of *Default* (in percent) calculated for a loan with a mean maturity of five quarters, and other mean characteristics, based on the coefficients of Model IV in Table 4. On the horizontal axes are the *Policy Rate prior to Origination* (INTEREST RATE_{t-1}) and the *Policy Rate until Maturity* (INTEREST RATE_{t+T-1}).



TABLE 1. DESCRIPTIVE STATISTICS

The table defines the variables employed in the empirical specifications, including their unit, and provides their mean, standard deviation, minimum and maximum. Panel A reports the dependent variables and Panel B the independent variables. All variables are measured in the quarter before the loan was granted, except the loan characteristics and the first three robust variables that are measured in the quarter the loan was granted. All country specific variables pertain to Spain unless otherwise indicated. EUR: amount in Euros, QTR: number of quarters, %: in percent. The number of observations equals 674,127 for all variables (based on a 3% random sample), except for Δ SMALL – LARGE BANK where it equals 13,265,830 (based on the full sample of firms with multiple loans).

<u>Panel A</u> Dependent Variables	Definition	Unit	Mean	St. Dev.	Min.	Max.
Probit						
RECENT BAD CREDIT HISTORY (0/1)	=1 if the borrower was overdue during the last six months on another loan; =0 otherwise. This information is known to the loan granting officer.	-	0.047	0.212	0	1
NO CREDIT HISTORY (0/1)	=1 if the borrower receives its first loan ever; =0 otherwise. This information is known to the loan granting officer.	-	0.053	0.224	0	1
Within Borrower Comparison						
Δ SMALL - LARGE BANK	The quarterly change in the difference between the percentages of the loan amounts that are obtained from either the borrower's small or large banks.	%	-0.084	34.170	-200.000	200.000
Duration Analysis						
DEFAULT (0/1)	=1 if there is default, i.e, if three months after the date of maturity or the date of an interest payment, the debt balance remains unpaid; =0 otherwise.		0.006	0.080	0	1
TIME TO DEFAULT	The number of quarters to default.	QRT	4.371	3.815	1	48

Panel B Independent Variables	Definition	Unit	Mean	St. Dev.	Min.	Max.
Monetary Policy Rates						
INTEREST RATE	Quarterly averages of German and Euro overnight interest rates (the Euro interest rate starts in 1999:I), dated in the quarter prior to loan origination.	%	4.135	2.166	2.023	9.619
∆ INTEREST RATE	The change in the INTEREST RATE	%	-0.037	0.702	-7.540	6.390
INTEREST RATE SPAIN	Quarterly average of Spanish overnight interest rate.	%	6.086	4.330	2.023	15.512
Bank Characteristics	including bank identity (213 banks)					
LN(TOTAL ASSETS)	The logarithm of the total assets of the bank.	EUR	16.859	1.626	8.595	19,484
LIQUIDITY RATIO	The amount of liquid assets (cash and balances with central banks, loans and advances to credit institutions, and loans and advances to general government) held by the bank over total assets.	%	29.270	12.422	0.222	95.902
BANK NPL _b -NPL	The difference between the bank and the average bank's level of non performing loans over total assets.	%	-0.013	1.793	-4.784	68.969
OWN FUNDS/TOTAL ASSETS	The amount of bank equity over total bank assets.	%	6.324	2.470	-11.226	80.945
LISTED (0/1)	=1 if the commercial bank is publicly listed; =0 otherwise.	-	0.456	0.498	0	1
SAVINGS BANK (0/1)	=1 if the bank is a saving bank; =0 otherwise.	-	0.319	0.466	0	1
CREDIT COOPERATIVE (0/1)	=1 if the bank is a credit cooperative; =0 otherwise.	-	0.050	0.218	0	1
Borrower Characteristics	including borrower identity (39,963 borrowers)					
LN(1+BORROWER BANK DEBT)	The logarithm of one plus the total amount of borrower bank debt.	EUR	5.609	2.380	0	14.401
LN(1+NUMBER OF BANK RELATIONSHIPS)	The logarithm of one plus the number of bank relationships of the borrower.	-	1.385	0.710	0	4.369
BAD CREDIT HISTORY (0/1)	=1 if the borrower was overdue any time before on another loan; =0 otherwise.	-	0.111	0.314	0	1
LN(2+AGE AS BORROWER)	The logarithm of two plus the age of the borrower. Age is the number of quarters from the first time the firm borrowed from a bank.	QRT	2.874	1.102	0.693	4.477
Loan Characteristics	including a loan identifier (674,127 loans)					
LN(SIZE OF THE LOAN)	The logarithm of the total loan amount granted.	-	4.175	1.376	1.792	15.061
COLLATERAL (0/1)	=1 if the loan is collateralized; =0 otherwise.	-	0.077	0.267	0	1
FINANCIAL CREDIT (0/1)	=1 if the loan is a financial credit; =0 otherwise. Financial credit includes all loans that are not used to finance either the production of commercialization of goods or services.	-	0.457	0.498	0	1
MATURITY 0m3m. (0/1)	=1 if the loan matures before 3 months; =0 otherwise.	-	0.421	0.494	0	1
MATURITY 3m1y. (0/1)	=1 if the loan matures between 3 months and 1 year; =0 otherwise.	-	0.375	0.484	0	1
MATURITY 1y3y. (0/1)	=1 if the loan matures between 1 year and 3 years; =0 otherwise.	-	0.099	0.298	0	1
MATURITY 3y5y. (0/1)	=1 if the loan matures between 3 year and 5 years; =0 otherwise.	-	0.035	0.185	0	1
Macro Controls						
GPDG	Growth in real gross domestic product.	%	3.032	1.312	-1.833	6.193
INFLATION	CPI Inflation rate.	%	3.755	1.368	1.490	7.100
COUNTRY RISK	The spread between the ten year Spanish and German government bond rate.	%	1.526	2.064	0.010	6.700
Province Characteristics						
NEW FIRMS/TOTAL FIRMS	The number of new firms over the total number of firms in the province.	%	4.551	1.530	1.148	21.781
HERFINDAHL-HIRSCHMAN INDEX	The Herfindahl Hirschman Index is hundred times the sum of squared bank market shares in total bank loans in each province (0 to 100).	-	6.950	3.001	2.578	39.420
Robustness						
CREDIT GROWTH	Growth of loans to firms at bank level.	%	1,922	30,636	-77,839	43.679
HOUSE PRICE GROWTH	The growth in house prices.	%	11 449	7 370	-7 415	27 335
FORECAST EURO AREA GDPG	ECB forecast for 1 year ahead euro area GDP growth.	%	2,210	0.526	1.200	3,400
SPANISH TEN YEAR BOND	The interest rate on a Spanish ten year Governement bond	%	6.877	3.483	3.180	14.750

TABLE 2. PROBIT MODELS

The estimates this table lists are based on probit models. The dependent variable is indicated in the table. In Panel A RECENT BAD CREDIT HISTORY equals one if the borrower was overdue during the last six months on another loan and equals zero otherwise. NO CREDIT HISTORY equals one if the borrower never received another recorded loan before and equals zero otherwise. In Panel B INTEREST RATE SPAIN is the quarterly average of the Spanish overnight interest rate. The definition of the other variables can be found in Table 1. Subscripts indicate the time of measurement of each variable. It is the quarter the loan was granted. Coefficients are listed in the first column, significance levels in the second column, and robust standard errors that are corrected for clustering at the borrower level are reported in italics between parentheses in the third column. *** Significant at 1%, ** significant at 5%, * significant at 10%.

<u>Panel A</u> Model	l Probit	l Probit		ll IV Probit		lll obit		IV Probit	IV	V Probit	Pr	VI obit		
Dependent Variable	RECENT B	RECENT BAD CREDIT HISTORY _{ft} (0/1) C		-1998 IT BAD	RECE	NT BAD		NO	198	88-1998 NO	NO CREDIT HISTORY _{ft} (0/1)			
	CREDIT HISTOR			TORY _{ft} (0/1)	CREDIT HI	STORY _{ft} (0/1)	CREDIT	HISTORY _{ft} (0/1)	CREDIT H	HISTORY _{fr} (0/1)				
Independent Variables	Coeff. Sig.	(S.E.)	Coeff. S	ig. <i>(</i> S. <i>E.</i>)	Coeff. S	Sig. (S.E.)	Coeff.	Sig. (S.E.)	Coeff.	Sig. (S.E.)	Coeff. S	Sig. (S.E.)		
Monetary Policy Rates														
INTEREST RATE ₁₋₁	-0.017 **	(0.008)			-0.133 **	* (0.027)	-0.019	*** (0.027)		-0.066 **	* (0.017)		
INTEREST RATE SPAIN _{T-1}			-0.048 ***	(0.014)					-0.033	*** (0.008)				
Monetary Policy Rate and Bank Characteristics														
INTEREST RATE _{τ-1} * LN(TOTAL ASSETS _{bt} -1)					0.007 **	* (0.002)					0.003 **	* (0.001)		

Bank Characteristics													
LN(TOTAL ASSETS _{br-1})	-0.006	(0.005)	0.010 *	(0.006)	-0.041 ***	(0.010)	0.013 ***	(0.010)	0.022 ***	(0.004)	0.001	(0.00	
LIQUIDITY RATIO	-0.014	(0.074)	-0.241 ***	(0.086)	0.007	(0.074)	0.132 ***	(0.074)	0.207 ***	(0.054)	0.138 ***	(0.04	
BANK NPL _{bt-1} -NPL _{t-1}	0.009 ***	(0.002)	0.006 ***	(0.002)	0.009 ***	(0.002)	-0.006 ***	(0.002)	-0.004 **	(0.002)	-0.006 ***	(0.00	
OWN FUNDS/TOTAL ASSETS _{bt-1}	-1.124 ***	(0.257)	-0.962 ***	(0.305)	-1.118 ***	(0.260)	0.367 **	(0.260)	0.886 ***	(0.208)	0.372 **	(0.15	
LISTED _{b1-1} (0/1)	-0.043 ***	(0.016)	-0.005	(0.019)	-0.048 ***	(0.016)	0.018 *	(0.016)	0.003	(0.013)	0.017 *	(0.01	
SAVINGS BANK _{bt-1} (0/1)	0.008	(0.017)	0.043 **	(0.020)	0.008	(0.017)	0.249 ***	(0.017)	0.240 ***	(0.013)	0.249 ***	(0.01	
CREDIT COOPERATIVE _{bt-1} (0/1)	0.109 ***	(0.036)	0.121 **	(0.054)	0.087 **	(0.036)	0.235 ***	(0.036)	0.267 ***	(0.027)	0.229 ***	(0.01	
Borrower Characteristics													
LN(1+BORROWER BANK DEBT _{it-1})	0.002	(0.010)	0.033 **	(0.013)	0.002	(0.010)							
LN(1+NUMBER OF BANK RELATIONSHIPS _{it-1})	0.619 ***	(0.035)	0.627 ***	(0.042)	0.618 ***	(0.035)							
LN(2+AGE AS BORROWER _{jt-1})	0.025 *	(0.015)	0.045 **	(0.019)	0.025 *	(0.015)							
Loan Characteristics													
LN(SIZE OF THE LOAN _{It})	-0.090 ***	(0.006)	-0.100 ***	(0.007)	-0.090 ***	(0.006)	-0.186 ***	(0.006)	-0.230 ***	(0.005)	-0.186 ***	(0.00	
COLLATERAL _{IT} (0/1)	0.342 ***	(0.023)	0.314 ***	(0.026)	0.342 ***	(0.023)	0.230 ***	(0.023)	0.100 ***	(0.017)	0.230 ***	(0.01	
FINANCIAL CREDITI _r (0/1)	-0.030 **	(0.013)	0.006	(0.015)	-0.030 **	(0.013)	0.281 ***	(0.013)	0.233 ***	(0.009)	0.282 ***	(0.00	
MATURITY _{It} 0m3m. (0/1)	-0.118 ***	(0.027)	-0.158 ***	(0.032)	-0.116 ***	(0.027)	-0.512 ***	(0.027)	-0.565 ***	(0.020)	-0.512 ***	(0.01	
MATURITY _{It} 3m1y. (0/1)	-0.046 *	(0.026)	-0.068 **	(0.030)	-0.045 *	(0.026)	-0.379 ***	(0.026)	-0.408 ***	(0.019)	-0.379 ***	(0.01	
MATURITY _{It} 1y3y. (0/1)	-0.009	(0.027)	-0.019	(0.032)	-0.009	(0.027)	-0.311 ***	(0.027)	-0.325 ***	(0.020)	-0.311 ***	(0.01	
MATURITY _{It} 3y5y. (0/1)	0.038	(0.027)	0.068 *	(0.036)	0.040	(0.027)	-0.151 ***	(0.027)	-0.181 ***	(0.024)	-0.150 ***	(0.01	
Macro Controls													
GPDG _{τ-1}	-0.088 ***	(0.007)	-0.049 ***	(0.010)	-0.088 ***	(0.007)	-0.007 *	(0.007)	-0.034 ***	(0.007)	-0.008 **	(0.00	
INFLATION,	0.028 **	(0.011)	-0.028 *	(0.015)	0.028 **	(0.011)	0.000	(0.011)	0.000	(0.012)	0.000	(0.00	
COUNTRY RISK,	0.022 ***	(0.008)	0.011	(0.010)	0.022 ***	(0.008)	0.042 ***	(0.008)	0.053 ***	(0.006)	0.042 ***	(0.00	
TIME TREND	0.029 ***	(0.010)	0.306 ***	(0.042)	0.026 **	(0.011)	-0.006	(0.011)	-0.045 *	(0.024)	-0.007	(0.00	
TIME TREND ²	0.000 ***	(0.000)	-0.001 ***	(0.000)	0.000 ***	(0.000)	0.000	(0.000)	0.000	(0.000)	0.000	(0.00	
Other													
CONSTANT	-2.751 ***	(0.854)	-20.240 ***	(2.743)	-2.000 **	(0.876)	-0.089	(0.876)	3.038 *	(1.564)	0.184	(0.37	
Industry dummies (9)	yes		yes		yes		yes		yes		yes		
Province dummies (49)	yes		yes		yes		yes		yes		yes		
No. of Observations (Loans)	674,127	7	292,346	6	674,12	7	674,12	7	292,34	6	674,12	7	
Log pseudolikelihood	-106,82	1	-430,131		-106,82	1	-125,63	5	-419,90)1	-125,629		
χ [*] (p-value)	0.000		0.000		0.000		0.000		0.000		0.000		

<u>Panel B</u> Model	ا OL	S							
Sample Period Dependent Variable	INTEREST R	TE SPAIN	INTEREST RATE SPAIN						
	All Variable	s in Level	All Variables in	Differences					
Independent Variables	Coeff. Sig.	(S.E.)	Coeff. Sig.	(S.E.)					
Monetary Policy Rates									
INTEREST RATE,	0.921 ***	(0.110)	0.634 ***	(0.221)					
Macro Controls									
GPDG ₇	0.167 **	(0.064)	-0.129	(0.111)					
	-0.201	(0.194)	-0.070	(0.177)					
	1.287 ***	(0.105)	0.995 ***	(0.105)					
Other									
CONSTANT	0.780 *	(0.410)	-0.075	(0.070)					
No. of Observations (Quarters)	43		43	3					
Adjusted R-squared	0.9	7	0.77						

TABLE 3. WITHIN BORROWER COMPARISON

The estimates this table lists are based on ordinary least square models. The dependent variable is Δ SMALL – LARGE BANK BORROWING, which is the quarterly change (between t-1 and t) in the percentages of the loan amounts that are obtained by each individual borrower from either the borrower's small or large banks (small/large banks are below/above the quarter median in total assets). The definition of the other variables can be found in Table 1. Model I analyzes per quarter all borrowers that have multiple bank relationships. Model II studies per quarter all borrowers with at least one relationship with a small and one relationship with a large bank. Subscripts indicate the time of measurement of each variable. t is the quarter the loan was granted. Coefficients are listed in the first column, significance levels in the second column, and robust standard errors that are corrected for clustering at the borrower level are reported in italics between parentheses in the third column.

*** Significant at 1%, ** significant at 5%, * significant at 10%.

		I				
Model	Coeff	OLS Sig (S	SF)	Coeff	OLS Sia	(S.F.)
Monetary Policy Rates	00011	0.g. (0		00011	e.g.	(0.2.)
	-0.071 '	***	(0.010)	-0.010		(0.012)
INTEREST RATE ₁₋₁ * RECENT BAD CREDIT HISTORY (0/1) _{jt-1}	-0.038 *	**	(0.019)	-0.091	***	(0.022)
Borrower Characteristics						
RECENT BAD CREDIT HISTORY (0/1)it-1	-0.036		(0.105)	0.342	***	(0.119)
LN(1+BORROWER BANK DEBT _{jr-1})	0.396 '	***	(0.014)	0.435	***	(0.020)
Macro Controls						
GPDG ₇₋₁	0.064 *	***	(0.010)	0.056	***	(0.012)
TIME TREND	-0.057 *	***	(0.010)	-0.045	***	(0.012)
TIME TREND ²	0.000 *	***	(0.000)	0.000	**	(0.000)
Other						
CONSTANT	4.291 '	***	(0.820)	2.368	**	(0.991)
Borrower Fixed Effects (769,222)		yes			yes	
No. of Observations (Quarters)	13,	265,830		6	,844,38	7
R-squared		0.03			0.05	

TABLE 4. DURATION MODELS

The estimates in this table are based on maximum likelihood estimation of the proportional hazard model using the Weibull distribution as baseline hazard rate. The parameter $ln(\alpha)$ measures the degree of duration dependence. The dependent variable is the hazard rate. The definition of the other variables can be found in Table 1. Subscripts indicate the time of measurement of each variable. t is the quarter the loan was granted. *T* is the time to repayment or default of the loan. *t* which indexes the time over the life of the loan, replaces *T* in Models VII and VIII in the subscripts of INTEREST RATE, GDPG, INLATION and COUNTRY RISK. All estimates are adjusted for right censoring. Coefficients are listed in the first column, significance levels in the second column, and robust standard errors that are corrected for clustering at the borrower level are reported in italics between parentheses in the third column. *** Significant at 1%, ** significant at 5%, * significant at 10%.

	I	I			111		IV		V		VI		VI		VIII	
Model	Weibull		Weibu	11	Weibu	II	Weibu	ll I	Weibu	III	Weibu	III	Weib	ull	Weibu	II
	Bank		Bank		Borrower/	Loan	Borrow	ver	Borrower	Risk	Differen	ces	Time-Va	rying	Time-Var	ying
	Heterogen	Heterogeneity		istics	Characteristics		Heterogeneity		Interact	ions					Without T	rends
Independent Variables	Coeff. Sig.	Coeff. Sig. (S.E.)		(S.E.)	Coeff. Sig. (S.E.)		Coeff. Sig.	(S.E.)	Coeff. Sig.	(S.E.)	Coeff. Sig. (S.E.)		Coeff. Sig	. (S.E.)	Coeff. Sig.	(S.E.)
Monetary Policy Rates																
INTEREST RATE _{T-1}	-0.099 ***	(0.019)	-0.084 ***	(0.021)	-0.072 ***	(0.021)	-0.127 ***	(0.023)	-0.100 ***	(0.025)			-0.114 ***	(0.014)	-0.052 ***	(0.014)
INTEREST RATE _{T+T-1}	0.344 ***	(0.014)	0.350 ***	(0.015)	0.323 ***	(0.016)	0.293 ***	(0.017)	0.324 ***	(0.016)			0.044 **	(0.021)	0.191 ***	(0.019)
INTEREST RATE _{τ-1} * BAD CREDIT HISTORY _{jt-1} (0/1)									-0.093 ***	(0.021)						
INTEREST RATE _{τ-1} * LN(2+AGE AS BORROWER _{jτ-1})									0.021 ***	(0.006)						
Δ INTEREST RATE ₇₋₁											-0.601 ***	(0.067)				
Δ INTEREST RATE ₇₋₂											-0.152 ***	(0.058)				
Δ INTEREST RATE ₇₋₃											0.011	(0.063)				
INTEREST RATE $_{\tau+T-1}$ - INTEREST RATE $_{\tau-1}$											0.201 ***	(0.012)				

Bank Characteristics								
LN(TOTAL ASSETS _{bt-1})		0.018 (0.014)	0.022 (0.015)	0.017 (0.015)	0.022 (0.015)	0.032 ** (0.015)	0.073 *** (0.015)	0.033 ** (0.014)
LIQUIDITY RATIO _{br-1}		-0.573 *** (0.215)	-0.358 (0.230)	-0.182 (0.218)	-0.362 (0.230)	-0.671 *** (0.233)	-1.804 *** (0.234)	-0.594 *** (0.198)
BANK NPL _{br-1} -NPL _{r-1}		0.061 *** (0.005)	0.055 *** (0.006)	0.069 *** (0.007)	0.055 *** (0.006)	0.053 *** (0.006)	0.051 *** (0.006)	0.055 *** (0.006)
OWN FUNDS/TOTAL ASSETS _{br-1}		-5.042 *** (0.922)	-5.360 *** (0.953)	-5.996 *** (0.922)	-5.449 *** (0.954)	-5.026 *** (0.948)	-5.543 *** (0.938)	-5.028 *** (0.928)
LISTED _{br-1} (0/1)		0.154 *** (0.059)	0.146 ** (0.060)	0.079 (0.060)	0.148 ** (0.060)	0.184 *** (0.060)	0.215 *** (0.062)	0.154 ** (0.060)
SAVINGS BANKhr-1 (0/1)		0.422 *** (0.058)	0.469 *** (0.059)	0.479 *** (0.059)	0.471 *** (0.059)	0.495 *** (0.059)	0.505 *** (0.061)	0.455 *** (0.059)
CREDIT COOPERATIVE		0.371 *** (0.099)	0.521 *** (0.103)	0.417 *** (0.105)	0.524 *** (0.103)	0.564 *** (0.103)	0.742 *** (0.104)	0.547 *** (0.101)
Borrower Characteristics								
LN(1+BORROWER BANK DEBT _{ir-1})			0.025 (0.018)	-0.043 ** (0.018)	0.023 (0.018)	0.027 (0.018)	0.014 (0.019)	0.019 (0.018)
LN(1+NUMBER OF BANK RELATIONSHIPS _{it-1})			0.529 *** (0.070)	1.013 *** (0.065)	0.538 *** (0.070)	0.540 *** (0.071)	0.540 *** (0.072)	0.537 *** (0.071)
BAD CREDIT HISTORY			1.110 *** (0.063)	0.777 *** (0.061)	1.608 *** (0.136)	1.132 *** (0.064)	1.093 *** (0.064)	1.118 *** (0.064)
LN(2+AGE AS BORROWER _{jt-1})			-0.406 *** (0.026)	-0.264 *** (0.027)	-0.512 *** (0.039)	-0.411 *** (0.026)	-0.367 *** (0.028)	-0.392 *** (0.027)
Loan Characteristics								
LN(SIZE OF THE LOAN,)			-0.231 *** (0.016)	-0.104 *** (0.018)	-0.235 *** (0.016)	-0.234 *** (0.016)	-0.238 *** (0.016)	-0.239 *** (0.016)
COLLATERAL _I (0/1)			0.225 *** (0.061)	0.076 (0.064)	0.227 *** (0.061)	0.240 *** (0.062)	0.237 *** (0.061)	0.227 *** (0.061)
FINANCIAL CREDIT _{IT} (0/1)			0.665 *** (0.056)	0.834 *** (0.055)	0.672 *** (0.056)	0.660 *** (0.056)	0.650 *** (0.057)	0.637 *** (0.057)
MATURITY _{Ir} 0m3m. (0/1)			0.906 *** (0.082)	1.182 *** (0.086)	0.912 *** (0.082)	0.976 *** (0.082)	0.950 *** (0.081)	0.947 *** (0.081)
MATURITY _{Ir} 3m1y. (0/1)			0.886 *** (0.068)	0.987 *** (0.071)	0.890 *** (0.068)	0.945 *** (0.069)	0.915 *** (0.068)	0.923 *** (0.068)
MATURITY _{Ir} 1y3y. (0/1)			0.527 *** (0.070)	0.495 *** (0.073)	0.530 *** (0.070)	0.565 *** (0.071)	0.520 *** (0.070)	0.549 *** (0.069)
MATURITY _{IT} 3y5y. (0/1)			0.158 ** (0.075)	0.111 (0.080)	0.159 ** (0.075)	0.173 ** (0.076)	0.139 * (0.076)	0.164 ** (0.075)
Macro Controls								
GPDG _{r-1}	-0.201 *** (0.014)	-0.202 *** (0.016)	-0.199 *** (0.017)	-0.240 *** (0.016)	-0.201 *** (0.017)		-0.162 *** (0.014)	-0.154 *** (0.014)
GPDG _{r+T-1}	-0.045 *** (0.016)	-0.048 *** (0.018)	-0.052 *** (0.018)	-0.057 *** (0.019)	0.227 *** (0.061)		-0.121 *** 0.017	-0.040 ** (0.016)
∆GPDG _{r-1}						-0.296 *** (0.047)		
GPDG _{r+T-1} - GPDG _{r-1}						0.026 * (0.015)		
INFLATION	0.067 ** (0.027)	0.055 * (0.031)	0.050 (0.031)	0.088 *** (0.030)	0.049 (0.031)	0.412 *** (0.033)	0.036 (0.029)	-0.094 *** (0.027)
	0.090 *** (0.021)	0.107 *** (0.023)	0.108 *** (0.023)	0.139 *** (0.023)	0.109 *** (0.023)	-0.058 ** (0.022)	0.033 (0.023)	0.222 *** (0.018)
TIME TREND	0.133 *** (0.021)	0.173 *** (0.028)	0.159 *** (0.028)	0.211 *** (0.026)	0.155 *** (0.028)	0.181 *** (0.023)	0.007 (0.024)	
TIME TREND ²	0.000 *** (0.000)	-0.001 *** (0.000)	0.000 *** (0.000)	-0.001 *** (0.000)	0.000 *** (0.000)	-0.001 *** (0.000)	0.000 * (0.000)	
Other	1.859							
CONSTANT	-19.04 *** (1.770)	-22.13 *** (2.276)	-21.89 *** (2.336)	-26.43 *** (2.158)	-21.48 *** (2.351)	-22.55 *** (1.839)	-5.684 *** (1.981)	-8.649 *** (0.430)
$ln(\alpha)$ (duration dependence)	0.620 *** (0.008)	0.619 *** (0.008)	0.671 *** (0.008)	0.816 *** (0.009)	0.672 *** (0.008)	0.624 *** (0.008)	0.705 *** (0.009)	0.671 *** (0.008)
Industry dummies (9)	no	no	yes	yes	yes	yes	yes	yes
Province dummies (49)	no	no	yes	yes	yes	yes	yes	yes
Year dummies (18)	no	no	no	yes	no	no	no	no
Bank Random Effects (213 Banks)	yes	no						
Borrower Random Effects (39,963 Borrowers)	no	no	no	yes	no	no	no	no
No. of Observations (Loans or Loan Quarters)	674,127	674,127	674,127	674,127	674,127	674,127	1,989,170	1,989,170
Log pseudolikelihood	-21,895	-21,682	-20,598	-19,199	-20,580	-20,842	-20,599	-20,719
χ ⁻ (p-value)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000

TABLE 5. DURATION MODELS INCLUDING INTERACTIONS WITH BANK CHARACTERISTICS

The estimates in this table are based on maximum likelihood estimation of the proportional hazard model using the Weibull distribution as baseline hazard rate. The parameter $ln(\alpha)$ measures the degree of duration dependence. The dependent variable is the hazard rate. The definition of the other variables can be found in Table 1. Subscripts indicate the time of measurement of each variable. t is the quarter the loan was granted. *T* is the time to repayment or default of the loan. All estimates are adjusted for right censoring. Coefficients are listed in the first column, significance levels in the second column, and robust standard errors that are corrected for clustering at the borrower level are reported in italics between parentheses in the third column. *** Significant at 1%, ** significant at 5%, * significant at 10%.

							IV		V		
Model	Weibul		Weib	ull	W	Veibull	Weibu	11	Weibu	11	
	Size		Liqui	lity		NPL	Type of Ow	nership	Local Ma	kets	
Independent Variables	Coeff. Sig.	(S.E.)	Coeff. Sig.	(S.E.)	Coeff.	Sig. (S.E.)	Coeff. Sig.	(S.E.)	Coeff. Sig.	(S.E.)	
Monetary Policy Rates											
INTEREST RATE _{τ-1}	-0.482 ***	(0.077)	-0.456 ***	(0.078)	-0.462 *	*** (0.079)	-0.563 ***	(0.090)	-0.412 ***	(0.082)	
INTEREST RATE _{T+T-1}	0.322 ***	(0.016)	0.325 ***	(0.016)	0.324 *	*** (0.016)	0.332 ***	(0.016)	0.328 ***	(0.016)	
Monetary Policy Rate and Bank or Province Characteristics											
INTEREST RATE _{r-1} * LN(TOTAL ASSETS _{br-1})	0.025 ***	(0.005)	0.031 ***	(0.005)	0.031 *	*** (0.005)	0.032 ***	(0.006)	0.030 ***	(0.005)	
INTEREST RATE _{T-1} * LIQUIDITY RATIO _{bT-1}			-0.304 ***	(0.078)	-0.302 *	*** (0.078)	-0.271 ***	(0.078)	-0.311 ***	(0.078)	
INTEREST RATE _{T-1} * BANK NPL _{bt-1} -NPL _{t-1}					0.119	(0.295)	0.007	(0.025)			
INTEREST RATE t_{-1} * LISTED t_{-1} (0/1)							0.037	(0.025)			
INTEREST RATE $_{t-1}$ SAVINGS BANK _{bt-1} (0/1)							0.090	(0.024)			
INTEREST RATE $_{\tau_1}$ CREDIT COOPERATIVE _{bt-1} (0/1)							0.024	(0.047)	-1 300 ***	(0.451)	
INTEREST RATE * HEREINDAHL HIRSCHMAN INDEX									0.664 *	(0.343)	
Bank Characteristics									0.001	(0.0.0)	
LN(TOTAL ASSETS _{br-1})	-0.111 ***	(0.028)	-0.134 ***	(0.029)	-0.136 *	*** (0.029)	-0.142 ***	(0.032)	-0.126 ***	(0.029)	
	-0.357	(0.229)	1.267 ***	(0.459)	1.258 *	*** (0.458)	0.992 **	(0.456)	1.039 **	(0.463)	
BANK NPL _{br-1} -NPL _{r-1}	0.058 ***	(0.006)	0.057 ***	(0.006)	0.050 *	*** (0.019)	0.057 ***	(0.006)	0.065 ***	(0.006)	
OWN FUNDS/TOTAL ASSETS _{bt-1}	-5.819 ***	(0.983)	-5.760 ***	(0.984)	-5.745 *	*** (0.985)	-5.601 ***	(0.982)	-4.956 ***	(0.955)	
LISTED _{bt-1} (0/1)	0.126 **	(0.060)	0.125 **	(0.060)	0.126 *	** (0.060)	-0.063	(0.140)	0.138 **	(0.059)	
SAVINGS BANK _{bt-1} (0/1)	0.459 ***	(0.059)	0.477 ***	(0.059)	0.478 *	*** (0.059)	0.014	(0.133)	0.468 ***	(0.059)	
CREDIT COOPERATIVE _{bt-1} (0/1)	0.448 ***	(0.103)	0.462 ***	(0.104)	0.462 *	*** (0.104)	0.280	(0.223)	0.434 ***	(0.102)	
Borrower, Loan, and Macro Controls, and Constant	yes		yes	;		yes	yes		yes		
Province Characteristics											
NEW FIRMS/TOTAL FIRMS _{pr-1}									0.086 ***	(0.027)	
HERFINDAHL HIRSCHMAN INDEX _{pr-1}									-0.018	(0.017)	
Other											
$ln(\alpha)$ (duration dependence)	0.671 ***	(0.008)	0.671 ***	(0.008)	0.671 *	*** (0.008)	0.673 ***	(0.008)	0.669 ***	(0.008)	
Industry dummies (9)	yes		yes	;		yes	yes		yes		
Province dummies (49)	yes	yes				yes	yes		no		
No. of Observations (Loans)	674,127	674,127		674,127		74,127	674,12	7	674,127		
Log pseudolikelihood	-20,580		-20,5	70	-2	20,570	-20,55	8	-20,692		
χ ² (p-value)	0.000		0.00	0	(0.000	0.000	1	0.000		

TABLE 6. DURATION MODELS – ROBUSTNESS

The estimates in this table are based on maximum likelihood estimation of the proportional hazard model using the Weibull distribution as baseline hazard rate. The dependent variable is the hazard rate. The definition of the other variables can be found in Table 1. Subscripts indicate the time of measurement of each variable. t is the quarter the loan was granted. *T* is the time to repayment or default of the loan. None of the variables vary over time. All estimates are adjusted for right censoring. Coefficients are listed in the first column, significance levels in the second column, and robust standard errors that are corrected for clustering at the borrower level are reported in italics between parentheses in the third column. *** Significant at 1%, ** significant at 5%, * significant at 10%.

	I			11			III			v		٧			VI			VII			VIII	I	IX	
Model	We	ibull	C:	Weibu	ll .		Weibu	ull	We	ibull	W	eibull		V	Veibu	 th		Weibu	ull	W	eibull	- T-	Weibu	ll here
Independent Variables	Coeff Sid	y <= 1 fr 1 (SE)	Coeff	Sin	(SE)	Coeff	Sin	(S.F.)	Coeff Si	-1996 1 (SE)	Coeff	99-200 Sia (ло (S.F.)	Coeff	Sin Sin	(SE)	Coeff	Sig	(S.F.)	Coeff S	in (SE)	Coeff	Sin	(SE)
Monetary Policy Rates	00011.04	. (0.L.)	000	oig.	(0.2.)	00011	oig.	(0.2.)	00011. 01	. (0.2.)	000111	oig. (1	0.2./	0000	olg.	(0.2.)	00011	oig.	(0.2.)	00011. 0	.g. (0.2.)	00011	oig.	(0.2.)
INTEREST RATE	-0.104 ***	(0.029)	-0.081	***	(0.030)	-0.111	***	(0.026)			-0.178	** (0	0.071)	-0.081	***	(0.021)	-0.077	***	(0.021)	-0.197 *	** (0.072	-0.066	***	(0.023)
INTEREST RATE SPAIN									-0.098 ***	(0.019)														
INTEREST RATE T+T-1	0.320 ***	(0.022)	0.307	***	(0.024)	0.397	***	(0.018)			0.266	*** (0	0.052)	0.322	***	(0.016)	0.324	***	(0.016)	0.268 *	** (0.052	0.324	***	(0.016)
INTEREST RATE SPAIN									0.325 ***	(0.010)														
Bank, Borrower, and Loan Controls	y	es		yes			yes		у	es		yes			yes			yes			yes			
Macro Controls and Constant																								
Additional Variables																								
CREDIT GROWTH _{br}														0.399	***	(0.067)								
HOUSE PRICE GROWTH,																	-0.009	*	(0.005)					
FORECAST EURO AREA GDPG _T																				-0.121	(0.085)		
SPANISH TEN YEAR $BOND_{\tau-1}$																						-0.029		(0.035)
Other																								
$ln(\alpha)$ (duration dependence)	0.766 ***	(0.009)	0.597	***	(0.012)	0.645	***	(0.010)	0.731 ***	(0.010)	0.802	*** (1	0.017)	0.670	***	(0.008)	0.672	***	(0.008)	0.804 *	** (0.017,	0.671		(0.008)
Industry dummies (9)	У	es		yes			yes		у	es		yes			yes			yes			yes		yes	
Province dummies (49)	у	es		yes			yes		у	es		yes			yes			yes			yes		yes	
No. of Observations (Loans)	536	,571	1	234,33	38	2	292,34	46	292	,346	38	81,781	l	674,127		7	674,127		27	38	1,781		674,12	27
Log pseudolikelihood	-8,	398		-8,359	9		12,60	06	-12,319		-7,671		-20,575		-20,595		-7,574		-20,597		97			
χ ² (p-value)	0.0	000		0.000)		0.000		0.000		(0.000		0.000		0.000		0.000		0.000)		

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS¹

- 0721 CLAUDIA CANALS, XAVIER GABAIX, JOSEP M. VILARRUBIA AND DAVID WEINSTEIN: Trade patterns, trade balances and idiosyncratic shocks.
- 0722 MARTÍN VALLCORBA AND JAVIER DELGADO: Determinantes de la morosidad bancaria en una economía dolarizada. El caso uruguayo.
- 0723 ANTÓN NÁKOV AND ANDREA PESCATORI: Inflation-output gap trade-off with a dominant oil supplier.
- 0724 JUAN AYUSO, JUAN F. JIMENO AND ERNESTO VILLANUEVA: The effects of the introduction of tax incentives on retirement savings.
- 0725 DONATO MASCIANDARO, MARÍA J. NIETO AND HENRIETTE PRAST: Financial governance of banking supervision.
- 0726 LUIS GUTIÉRREZ DE ROZAS: Testing for competition in the Spanish banking industry: The Panzar-Rosse approach revisited.
- 0727 LUCÍA CUADRO SÁEZ, MARCEL FRATZSCHER AND CHRISTIAN THIMANN: The transmission of emerging market shocks to global equity markets.
- 0728 AGUSTÍN MARAVALL AND ANA DEL RÍO: Temporal aggregation, systematic sampling, and the Hodrick-Prescott filter.
- 0729 LUIS J. ÁLVAREZ: What do micro price data tell us on the validity of the New Keynesian Phillips Curve?
- 0730 ALFREDO MARTÍN-OLIVER AND VICENTE SALAS-FUMÁS: How do intangible assets create economic value? An application to banks.
- 0731 REBECA JIMÉNEZ-RODRÍGUEZ: The industrial impact of oil price shocks: Evidence from the industries of six OECD countries.
- 0732 PILAR CUADRADO, AITOR LACUESTA, JOSÉ MARÍA MARTÍNEZ AND EDUARDO PÉREZ: El futuro de la tasa de actividad española: un enfoque generacional.
- 0733 PALOMA ACEVEDO, ENRIQUE ALBEROLA AND CARMEN BROTO: Local debt expansion... vulnerability reduction? An assessment for six crises-prone countries.
- 0734 PEDRO ALBARRÁN, RAQUEL CARRASCO AND MAITE MARTÍNEZ-GRANADO: Inequality for wage earners and self-employed: Evidence from panel data.
- 0735 ANTÓN NÁKOV AND ANDREA PESCATORI: Oil and the Great Moderation.
- 0736 MICHIEL VAN LEUVENSTEIJN, JACOB A. BIKKER, ADRIAN VAN RIXTEL AND CHRISTOFFER KOK-SØRENSEN: A new approach to measuring competition in the loan markets of the euro area.
- 0737 MARIO GARCÍA-FERREIRA AND ERNESTO VILLANUEVA: Employment risk and household formation: Evidence from differences in firing costs.
- 0738 LAURA HOSPIDO: Modelling heterogeneity and dynamics in the volatility of individual wages.
- 0739 PALOMA LÓPEZ-GARCÍA, SERGIO PUENTE AND ÁNGEL LUIS GÓMEZ: Firm productivity dynamics in Spain.
- 0740 ALFREDO MARTÍN-OLIVER AND VICENTE SALAS-FUMÁS: The output and profit contribution of information technology and advertising investments in banks.
- 0741 ÓSCAR ARCE: Price determinacy under non-Ricardian fiscal strategies.
- 0801 ENRIQUE BENITO: Size, growth and bank dynamics.
- 0802 RICARDO GIMENO AND JOSÉ MANUEL MARQUÉS: Uncertainty and the price of risk in a nominal convergence process.
- 0803 ISABEL ARGIMÓN AND PABLO HERNÁNDEZ DE COS: Los determinantes de los saldos presupuestarios de las Comunidades Autónomas.
- 0804 OLYMPIA BOVER: Wealth inequality and household structure: US vs. Spain.
- 0805 JAVIER ANDRÉS, J. DAVID LÓPEZ-SALIDO AND EDWARD NELSON: Money and the natural rate of interest: structural estimates for the United States and the euro area.
- 0806 CARLOS THOMAS: Search frictions, real rigidities and inflation dynamics.
- 0807 MAXIMO CAMACHO AND GABRIEL PEREZ-QUIROS: Introducing the EURO-STING: Short Term INdicator of Euro Area Growth.
- 0808 RUBÉN SEGURA-CAYUELA AND JOSEP M. VILARRUBIA: The effect of foreign service on trade volumes and trade partners.
- 0809 AITOR ERCE: A structural model of sovereign debt issuance: assessing the role of financial factors.
- 0810 ALICIA GARCÍA-HERRERO AND JUAN M. RUIZ: Do trade and financial linkages foster business cycle synchronization in a small economy?

^{1.} Previously published Working Papers are listed in the Banco de España publications catalogue.

- 0811 RUBÉN SEGURA-CAYUELA AND JOSEP M. VILARRUBIA: Uncertainty and entry into export markets.
- 0812 CARMEN BROTO AND ESTHER RUIZ: Testing for conditional heteroscedasticity in the components of inflation.
- 0813 JUAN J. DOLADO, MARCEL JANSEN AND JUAN F. JIMENO: On the job search in a model with heterogeneous jobs and workers.
- 0814 SAMUEL BENTOLILA, JUAN J. DOLADO AND JUAN F. JIMENO: Does immigration affect the Phillips curve? Some evidence for Spain.
- 0815 ÓSCAR J. ARCE AND J. DAVID LÓPEZ-SALIDO: Housing bubbles.
- 0816 GABRIEL JIMÉNEZ, VICENTE SALAS-FUMÁS AND JESÚS SAURINA: Organizational distance and use of collateral for business loans.
- 0817 CARMEN BROTO, JAVIER DÍAZ-CASSOU AND AITOR ERCE-DOMÍNGUEZ: Measuring and explaining the volatility of capital flows towards emerging countries.
- 0818 CARLOS THOMAS AND FRANCESCO ZANETTI: Labor market reform and price stability: an application to the Euro Area.
- 0819 DAVID G. MAYES, MARÍA J. NIETO AND LARRY D. WALL: Multiple safety net regulators and agency problems in the EU: Is Prompt Corrective Action partly the solution?
- 0820 CARMEN MARTÍNEZ-CARRASCAL AND ANNALISA FERRANDO: The impact of financial position on investment: an analysis for non-financial corporations in the euro area.
- 0821 GABRIEL JIMÉNEZ, JOSÉ A. LÓPEZ AND JESÚS SAURINA: Empirical analysis of corporate credit lines.
- 0822 RAMÓN MARÍA-DOLORES: Exchange rate pass-through in new Member States and candidate countries of the EU.
- 0823 IGNACIO HERNANDO, MARÍA J. NIETO AND LARRY D. WALL: Determinants of domestic and cross-border bank acquisitions in the European Union.
- 0824 JAMES COSTAIN AND ANTÓN NÁKOV: Price adjustments in a general model of state-dependent pricing.
- 0825 ALFREDO MARTÍN-OLIVER, VICENTE SALAS-FUMÁS AND JESÚS SAURINA: Search cost and price dispersion in vertically related markets: the case of bank loans and deposits.
- 0826 CARMEN BROTO: Inflation targeting in Latin America: Empirical analysis using GARCH models.
- 0827 RAMÓN MARÍA-DOLORES AND JESÚS VAZQUEZ: Term structure and the estimated monetary policy rule in the eurozone.
- 0828 MICHIEL VAN LEUVENSTEIJN, CHRISTOFFER KOK SØRENSEN, JACOB A. BIKKER AND ADRIAN VAN RIXTEL: Impact of bank competition on the interest rate pass-through in the euro area.
- 0829 CRISTINA BARCELÓ: The impact of alternative imputation methods on the measurement of income and wealth: Evidence from the Spanish survey of household finances.
- 0830 JAVIER ANDRÉS AND ÓSCAR ARCE: Banking competition, housing prices and macroeconomic stability.
- 0831 JAMES COSTAIN AND ANTÓN NÁKOV: Dynamics of the price distribution in a general model of state-dependent pricing.
- 0832 JUAN A. ROJAS: Social security reform with imperfect substitution between less and more experienced workers.
- 0833 GABRIEL JIMÉNEZ, STEVEN ONGENA, JOSÉ LUIS PEYDRÓ AND JESÚS SAURINA: Hazardous times for monetary policy: What do twenty-three million bank loans say about the effects of monetary policy on credit risk-taking?

BANCO DE ESPAÑA Eurosistema Unidad de Publicaciones Alcalá, 522; 28027 Madrid Telephone +34 91 338 6363. Fax +34 91 338 6488 e-mail: publicaciones@bde.es www.bde.es